Contents lists available at ScienceDirect

Heliyon

journal homepage: www.cell.com/heliyon

Review article

A process-driven and need-oriented framework for review of technological contributions to disaster management

Umair Iqbal*, Pascal Perez, Johan Barthelemy

SMART Infrastructure Facility, University of Wollongong, Australia

A R T I C L E I N F O

Keywords: Disaster management

Framework

Disruptive technology

Disaster constraints

Computer vision

Process-driven

Need-oriented

Systematic review

ABSTRACT

An escalation in the frequency and intensity of natural disasters is observed over the last decade, forcing the community to develop innovative technological solutions to reduce disaster impact. The multidisciplinary nature of disaster management suggests the collaboration between different disciplines for an efficient outcome; however, any such collaborative framework is found lacking in the literature. A common taxonomy and interpretation of disaster management related constraints are critical to develop efficient technological solutions. This article proposes a process-driven and need-oriented framework to facilitate the review of technology based contributions in disaster management. The proposed framework aims to bring technological contributions and disaster management activities in a single frame to better classify and analyse the literature. A systematic review of benchmark disruptive technology based contributions to disaster management has been performed using the proposed framework. Furthermore, a set of basic requirements and constraints at each phase of a disaster management process have been proposed and cited literature has been analysed to highlight corresponding trends. Finally, the scope of computer vision in disaster management is explored and potential activities where computer vision can be used in the future are highlighted.

1. Introduction

A disaster is defined as an event that occurs over a short or long period of time, affecting the entire community or society and wreaking havoc on people, the environment, infrastructure, wildlife, and the economy [1, 2, 3]. Hazard, vulnerability, and society's failure to deal with the disaster utilising the available resources are often the causes of disasters [4, 5, 6]. Disasters are broadly categorised as natural or manmade based on their origins. Natural disasters (e.g., earthquakes, floods, wildfires, landslides, liquefications, volcanic eruptions, hurricanes, cyclones, tsunamis, tornadoes, pandemics, blizzards) are those caused by natural phenomena such as geophysical and hydrological processes [4, 7, 8, 9]. On the other hand, technological or man-instigated disasters (e.g., industrial accidents, transportation accidents, terrorist attacks, war, nuclear radiation, stampedes, social unrest, conflicts, oil spills, and fires) are caused by the direct or indirect intervention of humans [8, 9]. Another type of induced disasters is referred to as a cosmic disaster, which encompasses nuclear war, bioterrorism, and climate change [9, 10]. This article mainly focuses on natural disasters to restrict the scope and complexity of the performed review. A gradual increase in the occurrence and intensity of natural disasters is noted over the past few decades. According to Guha-Sapir [11], a fraction of the increase in occurrence is because of advances in reporting, communication and detection technologies. However, the increase in hydro-metrological disasters including floods and droughts is real, which is believed because of climate change and unplanned urbanization in disaster-prone regions [12, 13]. Increased intensity, urbanization, and population growth are the highlighted factors which increase the vulnerability and make natural disasters more damaging in terms of deaths and their effect on people [5, 6, 14]. From 2010 till 2021, a total of 4384 natural disasters occurred, which resulted in 0.5 million deaths, 1886 million affected and 1.89 billion US dollar damages [11]. The ratio of affected people is more in underdeveloped countries mainly because of lack of resources to deal with natural disasters [13]. These horrifying disaster statistics have raised concerns in the community forcing the development of innovative solutions and extension of existing methods to reduce the impact of disasters towards proactively protecting the community [15].

The occurrence of a natural disaster cannot be avoided; however, the response can be improved to reduce its impact on the community. Because of the unpredictive nature, only limited resources can be

* Corresponding author. *E-mail address:* ui010@uowmail.edu.au (U. Iqbal).

https://doi.org/10.1016/j.heliyon.2021.e08405

Received 6 June 2021; Received in revised form 26 August 2021; Accepted 11 November 2021





^{2405-8440/© 2021} The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

allocated in advance to deal with natural disasters. Disaster management is the institution that involves the systematic strategic planning and deployment of procedures towards reducing the impact of disasters by efficiently using the available limited resources [16]. It is a multidisciplinary field and involves the collaboration between people from a variety of domains such as environmental sciences, logistics, civil engineering and computer sciences. Therefore, it incorporates the contributions from multiple disciplines to improve the overall management process. In the context of disaster management, interpretation of natural disasters has changed over the years towards the vulnerability of people to a certain hazard [5] and multiple models have been developed for disaster risk reduction. Some common models of vulnerability and disaster risk reduction include the Pressure and Release (PAR) model [17], Hazards-of-place (HOP) model [18], Regions of Risk model [19], Integrated Assessment of Multi-Hazards model [20] and the UNISDR framework for disaster risk reduction [21]. Given the involvement of actors from various domains as evident from above-mentioned models, it is significant to have collaboration and interaction between involved groups to better understand the needs of the community and efficient management. However, Seaberg et al. [22] highlighted the lack of any such comprehensive framework to connect the actors towards efficient disaster management. In this context, a common taxonomy and interpretation of disaster management related requirements are essential to develop the understanding of the problem to be addressed by a contributor (an integrator or technology provider).

In an effort to cope with disasters, academics and practitioners are developing different technology-oriented solutions [23]. Emerging disruptive technologies including Artificial Intelligence (AI) /machine learning/deep learning [24, 25, 26, 27, 28, 29, 30, 31, 32, 33, 34, 35, 36, 37, 38, 39, 40, 41, 42, 43, 44, 45, 46], social media/Big Data/crowdsourced data [45, 47, 48, 49, 50, 51, 52, 53], computer vision [36, 43, 44, 51, 52, 54, 55, 56, 57, 58, 59, 60, 61, 62, 63, 64, 65, 66, 67, 68, 69, 70, 71], Information Communication Technology (ICT)/ mobile phone/Geographical Information System (GIS) [46, 72, 73, 74, 75, 76, 77, 78], virtual reality/augmented reality [79, 80, 81, 82] and robotics/Unmanned Aerial Vehicles (UAVs) [43, 57, 62, 71, 77, 83, 84, 85, 86, 87] are highlighted as potential tools that may significantly help in improving the disaster management practices. Recent technological advancements in computing hardware, communication infrastructure and cloud computing have made it possible to practically implement the technological solutions to real disaster situations. Computer vision and Intelligent Video Analytics (IVA) have emerged as trending and economical technologies to address complex real-world problems. There is the enormous potential of deploying state-of-the-art computer vision algorithms to address disaster management problems. A section at the end of the article presents the scope of computer vision in disaster management and highlights potential challenges.

A number of reviews already published related to the role of specific technologies (e.g., AI [88], ICT [2], Big Data [15, 89]) in the disaster management domain; however, no comprehensive literature entry was found where trending technologies are reviewed at the same time in the context of disaster management to demonstrate the patterns of technology shift over the years. Furthermore, the process-driven and need-oriented framework proposed in this article provides a unique perspective of literature where a common taxonomy is established between disaster management problems and technology providers.

This article proposes a framework for a process-driven and needoriented review of the literature to help the contributors in a systematic classification of technological contributions to disaster management. The aim of the proposed review framework is to categorize the literature in a two-dimensional structure: horizontal axis progressing from prevention phase towards recovery phase of disaster management, while vertical axis detailing needs of each phase, corresponding assessments to address the needs and common technologies used in performed assessments. The proposed framework will help researchers in identifying gaps and proposing novel solutions to unaddressed disaster management problems. Furthermore, to facilitate the process of requirement formulation, this article proposes a list of common constraints to be properly defined and incorporated in the development process. As a summary, the followings are the main contributions of this article:

- 1. Presents a systematic review of benchmark technological contributions across different phases of the disaster management cycle to demonstrate the trends of technology used over the last decade.
- 2. Proposes a process-driven and need-oriented framework for the analysis of literature to help contributors in a systematic classification of literature and identifying pinpoint gaps.
- Proposes a list of disaster management related constraints and requirements with the idea of facilitating contributors in better addressing the problems and positioning their solutions in line with these requirements.
- 4. Explores the scope of computer vision in addressing disaster management related assessments and highlights the potential challenges.

The rest of the article is organized as follows. Section 2 presents contextual information about the modern disaster management cycle. Section 3 presents the methodology and protocols adopted to perform the presented review. Section 4 presents the review of technological advancements across different phases of the disaster management cycle. Section 5 proposes the process-driven and need-oriented framework and analyses the cited literature using the proposed framework. Section 6 lists disaster management related constraints and analyses the cited literature against the defined requirements. Section 7 presents the scope of computer vision for disaster management and highlights the potential challenges in this context. Section 8 concludes the study and reports important insights from the presented review.

2. Disaster management cycle

There are a variety of disaster management frameworks and procedures proposed in literature [90] to deal with disaster situations. The modern disaster management cycle is one of the most commonly and widely used frameworks, which consists of four phases; prevention and mitigation, preparedness, response and recovery [13, 16, 22, 90, 91, 92, 93, 94]. Each phase of the disaster management cycle is briefly described as follows:

- **Prevention and Mitigation** phase aims to minimize the impact of unavoidable future disasters in the long term. It involves the activities such as risk analysis, hazard zone mapping, resources allocation, climate forecasting and building warning codes.
- **Preparedness** phase involves the planning of responding to a nearfuture disaster. It involves the activities of disaster training, disaster exercises, and early warning systems.
- **Response** phase aims to reduce the impact and damages caused by the disaster after its immediate occurrence. It involves the activities of disaster mapping, damage estimations, search and rescue missions, and humanitarian assistance.
- **Recovery** phase aims to bring the community back to normal after a disaster occurrence. It involves the activities of reconstruction monitoring, debris clearance, and financial assistance.

Fig. 1 shows the functional block diagram of the modern disaster management cycle.

3. Review methodology and protocols

The presented review is performed using standard Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) [95] guidelines with proposed process-driven and need-oriented analysis of literature. The protocol consisted of the following steps (a)





Fig. 1. Functional block diagram of modern disaster management cycle (reconstructed from [89, 90, 94]).

Table 1. List of search keywords.

"Disaster Management" AND "Phases"	"Disaster Management" AND "Cycle"
"Disaster Management" AND "Activities"	"Disaster Management" AND "Assessments"
"Disaster Management" AND "Dequirements"	"Disaster Management" AND "Constraints"
Disaster management AND Requirements	Disaster Management AND Constraints
"Disaster Management" AND "Technology"	"Disaster Management" AND "Solutions"

research questions formulation, (b) keywords selection, (c) academic databases selection, (d) inclusion/exclusion criteria, (e) descriptive analysis of selected literature, (f) evaluation of selected literature. To restrict the scope of the presented review, technological contributions are referred to literature entries where disruptive technologies (i.e., robotics, computer vision, machine learning, AI, Big Data, virtual reality, augmented reality, statistical/probabilistic modelling, remote sensing, UAVs, crowdsourcing, social media, ICT, Structure from Motion (SfM), Internet of Things (IoT), GIS) are deployed to address the disaster management related problems.

A number of research questions listed as follows were formulated to identify the needs involved, highlight a variety of assessments performed and explore technological contributions made across phases of the disaster management cycle.

- What are the different phases and needs involved at each phase of the disaster management cycle?
- What are different assessments and activities performed at each phase of the disaster management cycle to address the corresponding needs?
- What are different technological solutions proposed to facilitate disaster management related assessments?
- What are different disaster management related constraints and requirements associated with each phase?

A list of corresponding keywords presented in Table 1 was prepared to extract the relevant literature. Scopus, Web of Sciences (WoS) and IEEEXplore academic databases were searched against the defined keywords and literature was extracted. In total, 41946 records were extracted, 11769 from Scopus, 23905 from WoS and 6272 from IEE-EXplore. Extracted literature was filtered using detailed and specific inclusion/exclusion criteria listed as follows to restrict the scope of performed review to the only benchmark studies.

- Only peer-reviewed literature (i.e., articles published in impact factor journals) published between 2010 and 2021 is considered for this review.
- · Literature published only in the English language is considered.

• Literature involving only selected natural disasters (i.e., earthquake, avalanche, wildfire, drought, flood, tsunami, landslide, hurricane, heatwave, volcanic eruption, gully erosion) is included in the main review.

Duplicate entries were removed and the relevance of the article was assessed at stages, including title screening, abstract screening and fulltext evaluations. It is inevitable to avoid the biasness and subjectivity in the process of screening the literature for inclusion in the final review despite having detailed inclusion/exclusion criteria as mentioned by Galindo and Batta [96]. A subjective test of "if the article proposes some technology-oriented solution to address one of disaster management related assessments" was applied during the full article screening process to determine the relevance. As a result, 111 articles were selected to be included in the presented review (i.e., 102 for the main review, 09 for framework support). The literature chosen for inclusion in this review is by no means near to the amount of research published in this domain; however, it can be considered as a representative sample of the most recent state-of-the-art literature. Fig. 2 presents the standard PRISMA flow diagram.

Exploratory analysis were performed on selected literature to highlight the year-wise and phase-wise trends. Fig. 3(a) presents the yearwise distribution of literature and indicates that the use of technology in disaster management has been increased significantly from 2017 onward. Fig. 3(b) shows the distribution of literature across phases of the disaster management cycle. Prevention and response phases were targeted the most from a technology perspective, while the recovery phase was least explored.

4. Review of technological contributions to disaster management

This section presents the review of technology-oriented solutions proposed to address assessments at different phases of disaster management cycle. The review is presented in the chronological order to highlight the shift of technology over the years.

4.1. Prevention and mitigation

In the year 2010, Bai et al. [24] proposed the use of GIS based logistic regression for landslide susceptibility mapping. Four different data



Fig. 3. Exploratory analysis of selected literature.

types including remote sensing, thematic maps, topographical maps, and geological maps were used for the investigation. Performance of regression model based mapping was measured using Root Mean Squared Error (RMSE) and classification accuracy. The proposed logistic regression model achieved an RMSE of 0.392 and an accuracy of 81.4%. Later in the same year, Martyr et al. [97] used high-resolution and unstructured mesh computational models (i.e., SWAN + ADCIRC SL16) of the Mississippi river to simulate the hurricane storm surge under varied flow conditions. To parameterize the effect of ripples from low to high flows, velocity variations were implemented. Model accuracy was assessed based on the quadratic fit estimates and R^2 value of 0.93 was reported. Furthermore, it was reported that the implementation of boundary conditions improved the overall model performance.

In 2012, Aronica et al. [98] proposed the use of Monte Carlo simulation to generate flood hazard maps. The hydrodynamic model was used to derive the flood event characteristics in the performed simulation. In the same year, Sun et al. [72] discussed the non-structural measures for the mitigation of floods and landslide hazards. Eight nonstructural measures including safety zones, critical rainfalls, monitoring equipment, warning systems, GIS data, hierarchical responsibility system, guide, and training were discussed in this context. Campos et al. [99] proposed the use of transportation network to determine the independent paths from disaster struck area to safe locations by repeatedly applying an analytical heuristic function. Travelling time and capacity of transportation networks were used as the main parameters in the defined heuristic function. In 2014, Caballero and Rahman [100] deployed Monte Carlo stochastic simulation approach for design flood estimation. Data from two catchments in Australia was used as a case study. The advantage of simulation in selecting the appropriate input value for flood modelling was reported. Later in the same year, Radianti et al. [101] proposed a novel Spatio-temporal probabilistic model integrating crowd and hazard dynamics to simulate the crowd flow. Efficient crowd flow simulation can help in better evacuation planning during a disaster event. A Dynamic Bayesian Network (DBN) was used to simulate the flow.

In 2015, Liu et al. [102] proposed the use of a 1D-2D hydrodynamic model for flood simulation in detention basins. A novel approach was presented to link 1D and 2D models for flood simulation. Flood mapping was generation using GIS. Ramakrishnan et al. [103] in year 2016 proposed the use of phase change materials in the building fabrics towards reducing the risk during heatwave disaster events. A numerical simulation approach was adopted to investigate the potential of phase change material towards resisting heat stress risk. From the results, improved performance was reported when phase change material was used. In 2017, Zhu et al. [83] proposed the use of UAVs for efficient floods related baseline data collection. It was reported that UAV's captured images provided improved flooding related information when combined with geographic information. Chen et al. [25] in 2018 performed a comparative study using multiple machine learning regression approaches (i.e., Bayes' Net (BN), Radical Basis Function (RBF), Logistic Model Tree (LMT), Random Forest (RF)) for landslide susceptibility mapping. A landslide spatial dataset for a custom site was used in a 70:30 ratio for training and testing purposes. RF model was reported best with an accuracy of around 75%. In the same year, Tanaka et al. [104] proposed a requirement definition method to develop a global disaster database by incorporating data from multiple sources. Various critical issues and challenges were highlighted in developing a global database including data generality and validation. Zhang et al. [54] in the year 2018 proposed a novel Object-based Convolutional Neural Network (OCNN) for urban land use classification from Very Fine Spatial Resolution (VFSR) images. Instead of image pixels, segmented objects were used as functional units for CNN. Improved land use classification accuracy was reported when compared with existing literature.

In 2019, Bera et al. [73] proposed the use of a multi-criteria analysis approach with GIS for landslide hazard zone mapping. Multiple landslide related factors were integrated into the platform and risk scores were determined. The effectiveness of the proposed approach was validated using a case study. In the same year, Komolafe et al. [105] developed a loss function for flood risk assessment by incorporating multiple flood damage related features. The proposed models were validated using a case study and the effectiveness of loss function in precisely predicting the loss was reported. Later in 2019, Zhang et al. [55] proposed a novel joint deep learning framework for land use and land cover classification by making use of CNN and Multilayer Perceptron (MLP) approaches. Improved classification accuracy were reported for the proposed joint deep learning approach when applied to VFSR images. Furthermore, the advantage of using the iterative update method was highlighted.

Recently in 2020, Abdi [26] performed a comparative study using multiple machine learning algorithms (i.e., Support Vector Machine (SVM), RF, xgboost, deep learning) for land use and land cover classification. The sentinel-2 dataset was used for training the machine learning models. From the results, SVM was reported best in the classification of data. In the same year, Band et al. [27] proposed the use of hybrid and ensemble machine learning models for the prediction of flash flood prone areas and generating flood susceptibility maps. Fifteen features related to climate and geo-environment were used as input to models for training. Topographical and hydrological features were reported as the most significant in the final prediction. Chowdhuri et al. [28] in 2020 implemented state-of-the-art Artificial Intelligence ensemble models (i.e., Boosted Regression Tree (BRT), Bayesian Additive Regression Tree (BART), Support Vector Regression (SVR), Ensemble of SVR) for the gully erosion susceptibility mapping. From the results, the Ensemble approach was reported to achieve the highest accuracy.

In 2020, Harirchian and Lahmer [106] proposed a fuzzy logic model for seismic vulnerability assessment of buildings in the context of an earthquake disaster. A novel rapid visual screening framework was used for earthquake vulnerability assessment of buildings. The effectiveness of the proposed framework was demonstrated using two case studies. Later in the same year, Lestari et al. [107] tested a contingency plan for volcanic eruption by finding a communication model through performing in-field training exercise simulations. Data was collected using in-field interviews, questionnaires, and surveys. The collected data were qualitatively analysed and interpreted for an efficient contingency plan. Mboga et al. [56] in the year 2020 proposed a fully connected CNN for land cover classification from historical aerial photographs. Two CNN architectures (i.e., atrous layers without downsampling, upsampling layers with downsampling) were deployed and reported to achieve an accuracy of approximately 90% outperforming conventional machine learning approaches.

Mitsuhara and Shishibori [79] in the year 2020 developed a virtual reality and augmented reality based Simulated Tornado Experience (STE) for better preparation for future tornado events. Series of comparative experiments were performed and virtual reality based STE was reported better for controlling learner fear and system operations. In the same year, Pal et al. [108] used a similar approach to Chowdhuri et al. [28] and implemented ensemble machine learning approaches for gully erosion susceptibility mapping. From the results, Bagging ensemble was recommended for efficient gully erosion mapping. Pour et al. [109] proposed the use of Low Impact Development (LID) approaches along with conventional flood stormwater management to better mitigate the impacts of floods because of climate change. Advantages, limitations, challenges and potential future aspects of LIDs in context to mitigate the urban floods were investigated in detail.

Sansare and Mhaske [74] in the year 2020 proposed the use of a GIS software tool for flood and landslide hazard mapping. QGIS software tool helped in the identification of important flood and landslide related insights for future hazard mitigation. Shahri and Moud [29] proposed a novel Hybrid Block Based Neural Network (HBNN) model for landslide susceptibility mapping. The proposed model was the combination of the divide-and-conquer approach and genetic algorithm. The performance of the model was assessed using the Region of Convergence (ROC) curve and the area under the curve was reported as approximately 89%. Later in the 2020, Ziarh et al. [110] proposed a novel Multi-Criteria Decision Analysis (MCDA) based on the entropy and catastrophe theory for flood risk mapping. The functionality of the proposed approach was demonstrated using a case study. From the results, 93% accuracy in flood risk mapping was reported for the proposed approach.

Most recently in 2021, Arabameri et al. [30] proposed a novel ensemble of machine learning algorithms for predicting maps of gully erosion susceptibility. The genetic algorithm was combined with the xgboost machine learning model and deployed on a GIS database for gully erosion prediction. From the results, an accuracy of around 90% was reported for the proposed approach. In the same year, Ndehedehe et al. [31] assessed the variations in hydrological stores on drought intensities within Australia using machine learning regression approaches. Gaussian kernel based SVM was deployed as a regressor and Quantile Function Storage (QFS) for assessment. Very critical insights from analyses were reported to help drought resilience and efficient water resources management. Pal et al. [111] in 2021 proposed a spatial distribution based soil erosion map using empirical observations Revised Universal Soil Loss Equation (RUSLE) modelling. The importance of climate change and land cover variations was reported in controlling soil erosion. From the results, it was highlighted that if climate change and human activities are not controlled, by 2100, nearly 9.88% of land surface top soil will erode.

In one of the most recent studies, Rahman et al. [112] developed a machine learning and hydrodynamic modelling based flood hazard zone mapping system. The accuracy of the model was assessed using the ground truth values and was reported high with an R^2 score of 0.83. Recently in 2021, Roy et al. [32] proposed the use of multiple machine learning algorithms (i.e., SVM, Biogeography Based Optimization (BBO), Extremely Randomised Trees (ERT)) for the mapping of flood prone regions in the Ajoy River basin of India. Climate change and land-use planning related features were used as input to the machine learning models. From the results, BBO performed best among

the three for flood susceptibility mapping. In the same year, Saha et al. [33] investigated the soil erosion and land degradation phenomena by making use of machine learning and optimisation models. Specifically, Gully Erosion Susceptibility (GES) mapping was investigated, and land use planning was optimized by deploying RF, BRT and EBO models. From the results, Ecogeography based Optimization (EBO) was reported as the best model for GES mapping. Saha et al. [34] in the year 2021 proposed the use of ensemble machine learning models for flood susceptibility mapping. The ensemble of hyperpipes and SVM was implemented on eight floods related variables to predict the susceptibility maps. From the experiments, the proposed ensemble approach was reported to achieve an accuracy of 93%. Most recently, Shahri et al. [113] developed a 3D topographical model using a geospatial and data-driven algorithm for generating accurate subsurface topography spatial variations. Accuracy of 82% was reported from the results. Furthermore, it was demonstrated that the proposed model was helpful in determining the hydrological properties of the structure.

It has been observed that technology has facilitate variety of assessments and activities at prevention and mitigation phase including landslide susceptibility mapping, disaster simulations, flood hazard mapping, evacuation planning, crowd flow simulation, structural design modifications, baseline data collection, flood risk assessment, land-use and land-cover classification, gully erosion susceptibility mapping, seismic assessment of buildings, contingency planning, drought prediction and subsurface topography mapping. To assist above-mentioned assessments, technology has evolved from the conventional approaches (i.e., Regression, Computational Modelling, Monte Carlo Simulation, Probabilistic Modelling, Hydrodynamic Modelling, Conventional Machine Learning Models, Empirical Modelling, 3D Modelling, Fuzzy Logic, Mathematical Modelling) towards learning based and AI oriented approaches (i.e., GIS, UAVs, CNN, Deep Learning, Ensemble Machine Learning, Artificial Intelligence, Augmented Reality, Neural Networks, Remote Sensing). Furthermore, it has been observed that most of the studies are performed under a limited scope for a specific region.

4.2. Preparedness

In 2010, Chang et al. [35] proposed a clustering based hybrid inundation model using linear regression and Artificial Neural Networks (ANNs) for flood inundation forecasting. A simulated dataset using a two-dimensional non-inertial overland flow model was used for the training and testing of regression models. The proposed model was able to predict the flood inundation depths accurately 1 hour ahead. In the same year, Liao et al. [114] developed an early warning system for landslides by making use of geospatial and remote sensing data. The proposed system was capable of landslide susceptibility mapping, precipitation monitoring and landslide prediction. In 2011, Zhang et al. [115] developed a real-time landslide monitoring system using a wirepulling trigger displacement meter and grid pluviometer. Software and hardware combination was used to acquire data from sensors and forecasting of landslide on the highway.

Lin et al. [36] in 2013 proposed a flood forecasting model using a two-stage support vector machine. The idea of forecasting the rainfall from typhoon characteristics along with observed rainfall at the first stage and using forecasted rainfall along with observed runoff to forecast the final runoff at the second stage was proposed. From the results, the proposed models were able to forecast accurately up to 6 hours ahead. In the same year, Devi et al. [116] developed a disaster prediction system based on the statistical analysis. SPSS data mining tool was used to interpret the spatial data for disaster prediction. The proposed approach was reported effective in analysing the data and predicting the future disaster event. In the year 2014, Chang et al. [37] proposed the use of multiple ANN models (i.e., static ANN, Elman NN, Nonlinear Autoregressive Network with Exogenous inputs (NARX)) towards forecasting the water levels from the rainfall data. From the results, the NARX model was reported to accurately forecast the water levels at the selected station with 10-60 min ahead. Later in the same year, Dehghani et al. [38] deployed Neural Networks and Monte Carlo simulation approaches for drought forecasting. Streamflow discharge data was used and was indexed for being used in the simulations. From the analysis, forecasted results were reported within 95% of the confidence interval. Lucieer et al. [57] in the year 2014 proposed a cost-effective UAV based landslide monitoring system where structure from motion approach was used to map the landslide displacements using UAV captured imagery. The accuracy was validated using ground truth values and reported as 7 cm and 6 cm horizontal and vertical accuracy, respectively.

In 2015, Asharose et al. [117] proposed an education tool to conduct disaster awareness workshops to better prepare against hazards. Awareness workshops were reported helpful in enhancing the knowledge of people related to disasters and helped them in better dealing with disaster situations. Gong et al. [80] in year 2015 proposed the use of virtual reality technology for earthquake disaster awareness. A head mount hardware was used to simulate the disaster event in virtual to educate the community. From the results, the proposed technology proved effective in simulating earthquake events and improving disaster drills. In the same year, Moghari and Araghinejad [39] used multiple variants of neural networks for the forecasting of the monthly and seasonal drought. From the results, it was reported that forecasting accuracy increased with precipitation time-scale increase while decrease with an increase in lead time. In 2016, Hajian et al. [118] proposed a model to simulate wildfire propagation considering it a stochastic shortest path problem. Monte Carlo simulation approach was used to identify the distribution of fire travelling time. The proposed approach resulted in comparatively faster wildfire prediction with an acceptable effect on accuracy.

In 2017, Asim et al. [40] proposed the use of machine learning approaches (i.e., Recurrent Neural Network (RNN), NN, RF and Ensemble) to predict the magnitude of the earthquake in the Hindukush region. Models were trained using the mathematically calculated eight seismic indicators. From the analysis, encouraging results were reported. Later in the same year, Klise et al. [119] proposed Water Network Tool for Resilience (WNTR) software for water distribution resilience during an earthquake disaster event. Damage to the water distribution network was reported dependent on both the magnitude of the earthquake and the available resources to repair the network. In 2018, Hu et al. [81] proposed the use of virtual reality based simulations to help the community prepare for disasters. 3D reconstructed disaster situations were rendered using virtual reality in this context. From the analysis, it was reported that the proposed approach could help in better awareness of disaster situations. In year 2019, Seibert et al. [120] also proposed the use of citizen science for water level measurements by CrowdWater application. From the preliminary experiments, placement of virtual gauge size was found problematic.

In 2019, Berkhahn et al. [41] developed an ensemble of neural networks to predict the water levels in real-time during a flooding event. A new network growing algorithm was proposed to select the topology of the ensemble network. The dataset used for training and testing of models was created using HE 2d hydrodynamic model. Acceptable forecasting accuracy with reduced computation time was reported for the proposed approach. In the same year, Wang et al. [58] proposed dilated casual CNN for the water level forecasting with a 1-hour to 6-hour lead time. The proposed approach was validated on a diversity of data collected from multiple typhoons and benchmarked against the conventional machine learning models. In the year 2020, De Vitry and Leitao [59] investigated the proxy water level measurements' impact on the pluvial flood forecasting models' performance. Image-based proxy water level measurements were used and multiple calibration tests were performed to study the impact. Pluvial flood forecasting model performance was reported enhanced when calibrated using proxy water level measurements; however, the presence of complex correlated errors was also reported, which may have a negative impact on model performance.

Recently in 2020, Mishra et al. [75] proposed the use of GIS technology for monitoring water levels and forecasting floods. A satellite altimetry derived approach was deployed for water level monitoring of large water bodies. Furthermore, a web based front-end was developed to visualize the flood maps. In 2020, Strobl et al. [121] proposed the use of citizen science base mobile application CrowdWater for water level measurements. The virtual-gauge functionality of the CrowdWater application was validated against the in-field surveys and from the analysis, application recorded water levels were reported comparatively accurately to field values. In 2021, Pillai et al. [122] developed a service-oriented IoT architecture for disaster forecasting and early warning by using machine learning algorithms over the cloud server. Acceptable absolute errors were reported when machine learning models were applied on three different sensors data on the cloud. In one of the most recent studies, Tamakloe et al. [123] developed an algorithm based on the topological features and Spatio-temporal traffic concentration within the network for better evacuation during disaster situations. From the experimental results, the proposed approach performed better in comparison to the conventional shortest passage approach.

From the cited literature in this section, it is observed that technology has been used to facilitate variety of activities at preparedness phase including flood inundation forecasting, early warning systems, landslide monitoring system, flood forecasting, water level forecasting, drought forecasting, disaster awareness workshops, earthquake magnitude prediction, water distribution network resilience and efficient evacuation planning. To assist above-mentioned activities, different solutions were proposed using technologies including ANNs, remote sensing, data acquisition, machine learning regression, Monte Carlo simulation, UAVs, ensemble of machine learning, virtual reality, citizen science, hydrodynamic modelling, and GIS.

4.3. Response

In 2010, Poser and Dransch [124] investigated the idea of using the Volunteered Geographic Information (VGI) as an efficient source to facilitate the activities within disaster response and recovery phases. A case study was presented where VGI was used for post flood rapid damage estimations and achieved comparable results to the conventional hydraulic modelling based estimates. In 2011, Barrington et al. [125] evaluated the use of crowdsourced satellite images for providing post-disaster damage assessments. Authors investigated the existing initiatives in this regard and highlighted the future aspects of how crowdsourced remote sensing can be used as a tool for quick damage assessments. Later in the same year, Bengtsson et al. [76] made use of mobile phone network data to track the post-earthquake movement of people for efficient response related decisions. A case study was presented to demonstrate the proposed idea and reported that population movement patterns could significantly help disaster management agencies in executing effective plans. Choi and Lee [84] in the year 2011 proposed the use of UAVs for real-time and effective monitoring of disasters. A system was developed using multiple sensors to acquire the disaster site data and transmit it to the ground station for analysis and effective decision making.

In 2013, Aghamohammadi et al. [42] proposed a neural network based approach for the estimation of human loss as a result of an earthquake event. The model was trained using the Bam 2003 earthquake data and reported the high estimation performance. In the same year Kao et al. [60] proposed a computer vision based monitoring and warning system for high risk flood debris flow. Optical flow and object based frame to frame comparison approaches were deployed for debris monitoring. Conventional image processing techniques including background subtraction, spatial filtering and entropy determination were also implemented. Lin et al. [36] in the year 2013 proposed a dualcamera setup (i.e., wide-angle, speed dome) for the high-resolution visual monitoring of disaster. Feature matching and stitching approaches were used to render high-resolution visuals for efficient disaster analysis. Houston et al. [47] in year 2014 proposed a framework to use social media in facilitating disaster management activities. The proposed framework connected all the actors involved within the disaster management domain and highlighted social media means to connect for efficient disaster response. In 2015, Albuquerque et al. [48] proposed the use of VGI extracted from social media and authoritative information from sensors to identify disaster related useful information. The effectiveness of the proposed approach was demonstrated for a flood case study and reported the usefulness of the proposed hybrid information combination. In the same year, Boccardo et al. [85] investigated the use of UAVs for post-disaster information gathering and mapping. The advantages and limitations of the UAV platforms were highlighted in this context. Furthermore, a framework was proposed for the efficient deployment of UAVs.

In 2016, Jiang and Friedland [126] proposed a mono-temporal image processing approach to classify the post-hurricane debris and non-debris satellite images. Multivariate texture features were reported helpful in the efficient classification of debris from non-debris regions for post-disaster damage assessment. In the same year, Kim et al. [82] proposed the use of mobile augmented reality to facilitate the process of post-disaster structural damage assessments and safety analysis. A framework was introduced to support the adaptability of virtual reality technology and a prototype system was developed to demonstrate its effectiveness. Later, Koyama et al. [61] proposed an advanced gradient based optical flow estimation approach to estimate the Tsunami debris from Synthetic Aperture Radar (SAR) images. The improved performance was reported for the image processing based approach with acceptable error. Kryvasheyeu et al. [49] in the year 2016 proposed the use of social media activity as a tool to assess the damage caused by a disaster. The proposed approach was validated using a Hurricane case study and reported the strong relationship between Twitter activity in assessing the damage.

In 2017, Bejiga et al. [43] proposed the use of a UAV equipped with a camera to facilitate post avalanche search and rescue operations. CNN and SVM were used for classifying the objects of interest from UAV captured images. Pre-processing and post-processing steps helped in improving the detection rate and prediction performance, respectively. In the same year, Ghosh and Gosavi [127] proposed a semi-Markov model for post estimation of disaster restoration time and quantifying the disaster rate during the response phase. The use of dynamic programming was proposed for the determination of the response centre. Later, Kakooei and Baleghi [62] in 2017 proposed the use of images from multiple sources (i.e., satellite, airborne, UAV) for post-disaster damage assessment. Facade images captured by UAVs when fused with roof images from satellite provided better damage assessments. Conventional image processing approaches were implemented to process the images. Arnold et al. [86] in year 2018 proposed the use of autonomous flying robots for facilitating search and rescue operations during initial hours of a disaster event. Swarm of autonomous flying robots was simulated in this context and was reported effective in locating 90% of the disaster victims. The swarm was controlled by cooperative behaviour intelligence approach in the simulations. In the same year, Galbusera and Giannnopoulos [128] investigated the role of different disaster modelling approaches in the context of input-output economic models for disaster impact assessments. Based on the equilibrium theory and economic production theory, input-out economic model variants are most commonly used for post-disaster impact assessments. In the same year, Vetrivel et al. [63] investigated the structural damage assessment using computer vision and deep learning approaches. 3D point cloud features in combination with multiple kernel learning were used to improve the generalization problem and effective post-disaster damage assessment of structures.

In 2019, Ahmad et al. [44] proposed novel deep learning approaches for the identification and detection of passable roads in satellite and social media images after a flooding disaster. A comprehensive dataset provided by MediaEval 18 was used and fusion techniques were de-

ployed to address the challenge. The improved performance was reported over the existing state-of-the-art approaches by the authors. In the same year, Aljehani and Inoue [77] proposed the use of UAVs to generate post-disaster safe maps based on the scanning of area and movement of population. Mobile network information was used to track the movement of the population, while UAV captured visuals were used to scan the area. Later, Bhola et al. [64] in the year 2019 proposed the use of computer vision based validation data for urban flood inundation forecasting. Computer vision algorithms were deployed to find the water depths mainly from the heights of reference objects in the images. From the results, it was reported that additional computer vision based validation data could help in significantly improving flood inundation forecasting. Bird et al. [50] discussed the potential of social media (i.e., Facebook) for flood related information sharing. The case of the 2011 Queensland floods event was specifically discussed to demonstrate how Facebook groups helped in sharing the community related information during the disaster event.

Huang et al. [51] in 2019 proposed the fusion of visual and textual features for automatic labelling of social media posts related to disaster for efficient response. Deep learning models were implemented for training on the fused features for automatic labelling. The addition of visual features improved the accuracy in comparison to only textual features case. Later in the same year, Liang [65] proposed the use of deep learning models with the Bayesian optimization approach for post-disaster structural inspection of concrete bridges. Deep learning classification, detection and segmentation approaches were deployed to extract the structural failure related information for making final "major failure" or "no failure" decision. An accuracy of over 90% was reported for structural integrity classification using the proposed approach. Madichetty and Sridevi [52] in the year 2019 proposed a Stacking-based Ensemble using Statistical features and Information Words (SESIW) for detection of social media tweets related the disaster damage assessments. The main features used in the model included hashtags frequency, URLs, and user mentions. The proposed approach, when validated on baseline Twitter datasets, outperformed the conventional machine learning models. Ogie et al. [53] in the year 2019 proposed the use of crowdsourced social media data for flood mapping. Results and important insights from the PetaJakarta project were presented where social media was effectively implemented for flood maps generation and other disaster management related activities.

In 2020, Arif et al. [66] performed a comparative investigation using multiple deep learning models for disaster detection from social media images. A custom visual dataset was developed, and existing CNN models were implemented. From the results, it was reported that the VGG model was able to successfully detect the type of disaster from social media visual content. In the same year, Hao et al. [45] proposed the use of multimodal social media data (i.e., text, images) for the post-disaster damage assessments using machine learning algorithms. From the investigation, it was reported that by use of both image and text analysis, the proposed approach was able to achieve accurate damage assessment and provided more useful information about the disaster. Later, Majumder et al. [129] proposed a post-earthquake debris management system by using a mathematical modelling approach. A mathematical model was proposed for the optimal cost of transportation of debris and the selection of suitable dumping locations. In a recent study, Zhai and Peng [67] proposed the use of Google Street View (GSV) for posthurricane damage assessment by applying deep learning approaches. From the results, it was reported that GSV proved helpful in damage assessments when damage level was low, while remote sensing images provided better damage assessment for high damage levels. In one of the most recent studies, Tay et al. [68] proposed the use of SSAR images from satellite to assess the damage caused by the floods. Change detection approach and GIS technology were implemented for damage mapping.

At response stage of disaster management cycle, technological solutions were proposed in abundant involving both conventional and state-of-the-art approaches to address the variety of assessments including disaster response, damage assessments, victims tracking, disaster monitoring, disaster loss estimation, debris management, disaster information gathering, search and rescue operations, determining clear transport routes and flood inundation mapping. Some significant technologies used at this stage included VGI, crowdsourcing, remote sensing, mobile phone, UAVs, NN, conventional computer vision, social media, SAR, CNN, conventional machine learning, semi-Markov modelling, deep learning, ensemble of machine learning, and mathematical modelling.

4.4. Recovery

In 2014, Hoshi et al. [69] explored the use of satellite images for post-earthquake recovery monitoring. Satellite images were visually interpreted and validated against the results of field surveys to demonstrate the effectiveness of satellite imagery in assessing the recovery. In 2017, Baytiyeh [78] investigated the role of ICT in post-earthquake education delivery. The effectiveness of online learning and the use of technology was highlighted in reinstating education after a disaster event. In 2017, Chowdhury et al. [87] proposed a Continuous Approximation (CA) model to explore the potential of drones in providing logistics to disaster-struck areas. The proposed model was able to identify optimal locations for distribution and estimate the overall operational costs.

In 2017, Hashemi-Parast et al. [130] used information from multiple resources (i.e., statistics, images from field surveys, satellite images) for the assessment of post-earthquake urban reconstruction monitoring in Bam, Iran. From the results, it was reported that satellite images and photographs helped in better assessment of reconstruction monitoring. However, the interpretation of visuals was made manually. In the same year, Shiraki et al. [131] proposed a resilience engineering approach for road network recovery after an earthquake. Mathematical heuristics and analytical rule formation approaches were used in the proposed framework for effective reinstatement of the road network. Later, Yang and Qi [70] proposed a system for spatial-temporal dynamic vegetation monitoring after an earthquake event. Normalized vegetation index time series data was used for the analysis and cross-correlogram and spectral matching approaches were implemented. From the results, it was reported that the proposed approach was able to efficiently monitor the vegetation growth in the region.

In 2018, Barabadi and Ayele [132] proposed the use of different statistical and machine learning models for the prediction of infrastructure recovery rates after disasters. Historical data was used for the analysis. The proposed framework was divided into three scenarios based on the availability and type of data (i.e., missing data, heterogeneous data, homogeneous data). The proposed approach helped in selecting the correct model for recovery rate monitoring based on the scenario. In the same year, Contreras et al. [133] assessed the recovery process after the 2009 L'Aquila earthquake event by using a recovery index based on spatial indicators. From the results, it was reported that the recovery index was helpful in highlighting the spatial pattern in the recovery process and can be used to quantify the recovery progress. In a most recent study performed in 2020, Soulakellis et al. [71] used the Unmanned Aerial System (UAS) captured data for post-earthquake recovery monitoring. High-resolution images and geo-information approach were used for efficient structural recovery mapping. The SfM approach was used for creating 3D point clouds of structures. The proposed approach proved 97% accurate in demolish detection. In 2021, Fan et al. [46] proposed the idea of Disaster City Digital Twin by incorporating ICT and AI into the process of facilitating disaster management activities. Data collection, data analysis, decision making and networking were reported as the four main components of the proposed paradiagram.

It has been observed from cited literature that recovery phase been least addressed from technology perspective. Assessments addressed at recovery phase included recovery monitoring, restoration of education, logistics restoration, reconstruction monitoring, restoration of transport routes and vegetation growth monitoring. Some highlighted technologies used in literature to address these activities included remote sensing, ICT, Continuous Approximation (CA) modelling, mathematical heuristics, machine learning, statistical modelling, UAS, AI, SFM and UAVs.

5. Process-driven and need-oriented framework based analysis

This section presents the proposed process-driven and need-oriented framework based analysis of cited technological contributions to disaster management. The proposed process-driven and need-oriented framework is designed to orient the literature in a two-dimensional structure to better align it against the respective phase of disaster management and corresponding assessment it aims to address. The horizontal axis of the framework represents different phases of disaster management progressing from prevention towards recovery. On the other hand, the vertical axis lists the needs/tasks involved at each phase, respective assessments/activities performed to address the needs and common technologies used to assist the assessments in addressing the disaster needs. Functional illustration of the proposed process-driven and needoriented framework deployed for the cited literature is given in Fig. 4. Some literature entries (i.e., [134, 135, 136, 137, 138, 139, 140, 141]) were included to support the framework and were not part of the main review as mentioned in Section 3.

From the framework, it can be interpreted that at the prevention phase, the most addressed need is "Disaster Risk Analysis and Management", and the most addressed assessment is the "Disaster Hazard Zone and Susceptibility mapping". Machine learning and deep learning were observed to be most deployed technologies at the prevention phase. At the preparedness phase, "Disaster Forecasting and Prediction" was the most addressed need in literature and corresponding "Disaster Pre-Warning Systems and Prediction Models" was the most explored assessment. Similar to the prevention phase, machine learning, deep learning and computer vision were the most used technologies. At the response phase, "Disaster Damage Analysis" was the most addressed need while "Structural Damage Assessment" was the most investigated assessment. Computer vision, social media and remote sensing were observed as the leading technologies used at this phase. Finally, at the recovery phase, "Restoration to Normal" was the most addressed need while corresponding "Reconstruction Monitoring" assessment was most explored from a research perspective. Computer vision, remote sensing, and UAVs were observed as the highlighted technologies used at this phase. Overall, it can be observed that the recovery phase has been least explored from a technological point of view and offers a huge potential to technology providers for deploying technological solutions to assist the assessments.

Tables 2 to 5 present the detailed process-driven and need-oriented analysis of cited literature for prevention, preparedness, recovery and response phases, respectively. Each literature entry is investigated for addressed disaster need, addressed disaster assessment, disaster type and technology used to address the corresponding assessment.

6. Constraints and set of requirements

Each phase of disaster management involves a number of constraints and a set of basic requirements associated with specific needs and problems. For a proposed solution to be effective, it must minimally address these requirements; however, from cited literature, proper requirement formulation is found lacking. Carter [16] mentioned some basic requirements to be addressed for effective disaster management performance. Building on the idea of Carter [16] and bringing it together with the proposed need-oriented review; this article proposes a list of basic constraints and a set of requirements to help the contributors propose efficient solutions to disaster management related problems. Proposed constraints are generalized and applicable in the scope of all common disasters; however, more comprehensive and assessment related constraints are recommended to be defined depending on the type of disaster and phase of the disaster management cycle. Listed are the proposed constraints:

- **Responsiveness** is referred to the speed of the proposed solution and is often measured by analyzing the processing times in providing the desired output. This constraint is closely related to the real-time functionality of the proposed approach and varies across different phases. In general, any proposed solution should process information as quickly as possible; however, it is not always the priority. Other than the assessments which require real-time operational functionality (e.g., flood monitoring, early warning systems, rapid flood mapping, search and rescue), responsiveness is not the priority in disaster management. Having said that, a problemspecific formulation of required response time is recommended.
- Accuracy is referred to the correctness of the proposed solution and is measured by validating against ground truth samples. In general, any proposed solution should be as accurate as possible; however, not always the priority. For the assessments where quick results are required (e.g., early disaster mapping, search and rescue, early visual inspections), accuracy is not as important as responsiveness. On the other hand, assessments where relaxed time constraints are involved (e.g., structural damage assessment, prevention measures, risk assessment, hazard zone mapping) should be highly accurate. For the assessment like early disaster warning systems, both accuracy and responsiveness are equally important, which makes it a challenging task for contributors to achieve. A specific formulation of accuracy requirement for the addressed problem is recommended for efficient outcome.
- Generalization is referred to the implementation diversity of the proposed solution and is assessed by validating the proposed method for comprehensive data containing multiple scenarios. In general, any proposed solution should be capable of performing under variable disaster situations without major changes. Having said that, it is not practical for most disaster management related assessments to have a highly generalized single solution applicable to all disasters. Therefore, it is highly recommended to properly formulate the generalization requirement specific to disaster type and assessment for efficient results.
- **Cost and Hardware Implementation** is referred to the overall cost and hardware implementation related constraints for the proposed solution. A feasibility study of the required budget, availability of hardware components, availability of power on-site, and communication network is highly recommended but often ignored by contributors. Logistic requirements vary significantly based on the type of disaster and assessment; therefore, an assessment-specific formulation of cost and implementation related requirements is recommended.
- User-Friendliness is referred to the ease by which a non-technical person can deploy, supervise and monitor a technological solution. It is highly desirable for a proposed system to be easy to operate by a non-technical individual with little knowledge. In practice, it is recommended to have a system that can be operated by anyone after a quick training session.

The cited literature has been evaluated against the proposed set of requirements to highlight the trends of addressed requirements. For this article, subjective evaluation criteria are defined and listed as follows:

- Responsiveness (I): If processing times for the proposed technological solution are minimally discussed.
- Accuracy (✓): If the performance of the proposed model is quantified by at least one measure.

	Pre-Disast	Disaster O	ster Occurrence Post-Disaster Phase						
			~		~				
Phases	Prevention and Mitigation	Preparedness		Response		Recovery			
Needs and Tasks	Disaster Risk Analysis and Management [83, 97, 99–102, 104] [26, 29, 31, 54–56, 79, 107, 109] Disaster Vulnerability Mapping [24, 25, 27, 28, 73, 98] [29, 30, 32–34, 74, 108, 110–112] Disaster Assessments [105, 106] Socio-Economic Analysis [134, 135] Disaster Prevention Measures [72, 103]	Disaster Awareness Programs [80, 81, 117] Disaster Forecasting and Prediction [35–37, 114–116] [38–41, 58, 59, 75, 118, 122] Pre-Disaster Monitoring of Vulnerable Regions [57, 119–121] Disaster Evacuation Planning [123]		Disaster Scope and Mapping [5, 53, 66, 77] Disaster Damage Analysis [5, 42, 82, 124–126] [45, 49, 52, 62, 63, 65, 67, 68] Disaster Economic Analysis [128] Disaster Information Gathering [48, 50, 51, 64, 76, 85] Disaster Vonitoring [36, 84, 127] Disaster Vonitoring [36, 84, 127] Disaster Debris Management [60, 61, 129] Disaster Response [44, 47]		Restoration to Normal [46, 69, 70, 130] Recovery Planning [71, 132, 133] Restoration of Transport [131] Re-Location of Community [136, 137] Restoration of Essential Services [78] Restoration of Goods Flow [87]			
Assessments and Activities	 Building Safety and Fire Codes [138] Land-Use and Land-Cover Classification [26, 54–56] Disaster Hazard Zone and Susceptibility Mapping [24, 25, 27, 28, 73, 98, 108] [29, 30, 32–34, 74, 110–112] Baseline Data Collection [83, 104] Efficient Evacuation Planning [99, 101] Structural and Non-Structural Mitigation Measures [72, 103] Disaster Simulation and Modelling [29, 31, 97, 100, 102] Disaster Damage Estimation [105] Contingency Planing [106] Disaster Mitigation Measures [109] 	Disaster Workshops and Training [80, 81, 117] Disaster Pre-Warning Systems and Prediction Models [35–37, 114–116] [38–40, 59, 75, 118, 122, 139] Water Level Measurements [41, 58, 120, 121] Disaster Resilience Monitoring [57, 119] Traffic and Routes Management for Effective Evacuation [123]		Rapid Damage Mapping [49, 68, 124, 125] Rapid Building Usability Assessment [140] Structural Damage Assessments [45, 52, 62, 63, 65, 67, 82] Search and Rescue Operations [43, 86] Disaster Impact and Economic Assessment [42, 127] Debris Detection and Estimation [60, 61, 126, 129] Disaster Surveillance and Mapping [36, 48, 76, 84, 85] Disaster Safe Zone and Area Mapping [35, 64, 77] Disaster Communication and Information Sharing [50, 51]		Reconstruction Monitoring [69,71,130,132,133] Vegetation Growth Monitoring [70] Restoring Education and Learning [78] Restoring Logistic Transport Network [87,131] Identification of Lessons Learnt [141] Disaster Recovery Framework [46]			
Technologies Used	Novel structural designs and research to improve resilience against disasters [103] Use of GIS and Machine Learning for disaster hazard zone mapping [24, 25, 27–29, 73, 74, 108] [30, 32, 33, 110, 112] Use of probabilistic modelling for effective evacuation plans [99, 101] Use of Virtual Reality, Augmented Reality and simulations for disaster risk analysis [29, 31, 79, 97, 100, 102] Use of CNN and deep learning for land-use and land-cover classification [26, 54–56] Use of UAVs for baseline data collection [83] Low Impact Development approach for disaster mitigation [109] Use of fuzzy logic modelling for seismic assessment of buildings [106]	Use of machine learning, deep learning and neural networks for disaster forecasting and prediction [35–40,122] Use of citizen science, IoT and SfM for disaster monitoring [57,120–122] Use of Monte Carlo simulation, GIS and mathematical modelling for disaster prediction [38,75,114,118] Use of educational tools and equipment for disaster awareness workshops [117] Use of virtual Reality for better disaster awareness [80,81] Use of sensors and data acquisition for disaster early warning systems [59,115] Use of computer vision for water level measurement [58] Use of statistical analysis for future disaster prediction [116]		Use of computer vision on satellite and airborne images for disaster mapping [64] Use of social media and crowdsourced data for disaster monitoring and response [47] Use of drones for disaster response related activities [43,62,77,84,85] Use of robotics technology for disaster search & rescue operations [43,86] Use of computer vision, GIS, VGI, remote sensing and social media for post-disaster damage assessments [45,49,52,62,63,65,68,124,125] Use of mobile phones and social media for disaster information gathering [48,50,51,76] Use of computer vision and mathematical modelling for disaster debris management [60,61,129]		Use of computer vision, satellite images, AI, SfM and UAVs for post-disaster reconstruction monitoring [69, 71, 130, 132, 133] ICT and online learning for post-disaster learning services [78] Use of resilience engineering and heuristic functions for transport routes restoration [131] Use of spectral matching and image processing techniques for vegetation growth monitoring [70] Use of Artificial Intelligence and ICT for Disaster Recovery [46]			

Fig. 4. Functional illustration of the proposed process-driven and need-oriented review framework.

- Cost and Hardware Implementation (✓): If hardware requirements and/or cost of deploying the proposed solution in practice is minimally addressed.
- User-Friendliness (✓): If operational information and user interface for the proposed solution are minimally addressed.
- Generalization (✓): If the proposed solution is validated against diversity of scenarios/data.

Tables 6 to 9 present the proposed constraint based analysis of the cited literature from the prevention, preparedness, recovery and re-

Table 2. The proposed process-driven and need-oriented analysis of cited literature from prevention and mitigation phase.

			F F		
Author/s	Year	Need and Tasks	Assessments and Activities	Disaster	Technology Used
Bai et al. [24]	2010	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Landslide	Conventional Machine Learning
Martyr et al. [97]	2010	Disaster Risk Analysis and Management	Disaster Simulation and Modelling	Hurricane	Computational Mathematical Models
Aronica et al. [98]	2012	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Flood	Monte Carlo Simulation
Sun et al. [72]	2012	Disaster Prevention Measures	Structural and Non-Structural Measures	Flood and Landslide	GIS and Equipment
Campos et al. [99]	2012	Disaster Risk Analysis and Management	Effective Evacuation Plans	All/Common	Analytical Heuristic
Caballero and Rahman [100]	2014	Disaster Risk Analysis and Management	Disaster Simulation and Modelling	Flood	Monte Carlo Simulation
Radianti et al. [101]	2014	Disaster Risk Analysis and Management	Effective Evacuation Plans	All/Common	Spatio-Temporal Probabilistic Model
Liu et al. [102]	2015	Disaster Risk Analysis and Management	Disaster Simulation and Modelling	Flood	1D-2D Hydrodynamic Models
Ramakrishnan et al. [103]	2016	Disaster Prevention Measures	Structural and Non-Structural Measures	Extreme Heat Wave	Phase Change Material, Simulations
Zhu et al. [83]	2017	Disaster Risk Analysis and Management	Baseline Data Collection	Flood	Unmanned Aerial Vehicles (UAVs)
Chen et al. [25]	2018	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Landslide	Conventional Machine Learning
Tanaka et al. [104]	2018	Disaster Risk Analysis and Management	Baseline Data Collection	All/Common	Database Development
Zhang et al. [54]	2018	Disaster Risk Analysis and Management	Land-Use and Land-Cover Classification	All/Common	Object-Based CNN (OCNN)
Bera et al. [73]	2019	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Landslide	GIS and Multi-Criteria Analysis
Komolafe et al. [105]	2019	Disaster Assessments	Disaster Damage Estimation	Flood	Mathematical Modelling
Zhang et al. [55]	2019	Disaster Risk Analysis and Management	Land-Use and Land-Cover Classification	All/Common	CNN and MLP
Abdi [26]	2020	Disaster Risk Analysis and Management	Land-Use and Land-Cover Classification	All/Common	Conventional Machine Learning
Band et al. [27]	2020	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Flood	Hybrid and Ensemble Machine Learning
Chowdhuri et al. [28]	2020	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Gully Erosion	Conventional Machine Learning
Harirchian and Lahmer [106]	2020	Disaster Assessments	Vulnerability Assessment of Buildings	Earthquake	Fuzzy Logic Model
Lestari et al. [107]	2020	Disaster Risk Analysis and Management	Contingency Planing	Volcanic Eruption	Communication Model and Training
Mboga et al. [56]	2020	Disaster Risk Analysis and Management	Land-Use and Land-Cover Classification	All/Common	Fully Connected CNN
Mitsuhara and Shishibori [79]	2020	Disaster Risk Analysis and Management	Disaster Simulation and Modelling	Tornado	Virtual Reality and Augmented Reality
Pal et al. [108]	2020	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Gully Erosion	Ensemble of Machine Learning
Pour et al. [109]	2020	Disaster Risk Analysis and Management	Disaster Mitigation Measures	Flood	Low Impact Development (LID) Approach
Sansare and Mhaske [74]	2020	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Flood	Geographic Information System (GIS)
Shahri and Moud [29]	2020	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Landslid	Novel Hybrid Neural Networks
Ziarh et al. [110]	2020	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Flood	Multi-Criteria Decision Analysis
Arabameri et al. [30]	2021	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Gully Erosion	Ensemble of Machine Learning
Ndehedehe et al. [31]	2021	Disaster Risk Analysis and Management	Disaster Simulations and Modelling	Drought	Conventional Machine Learning
Pal et al. [111]	2021	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Soil Erosion	Empirical Modelling
Rahman et al. [112]	2021	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Flood	Hydrodynamic Models and Machine Learning
Roy et al. [32]	2021	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Flood	Machine Learning Models
Saha et al. [33]	2021	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Soil Erosion	Machine Learning and Optimization Models
Saha et al. [34]	2021	Disaster Vulnerability Mapping	Disaster Hazard Zone and Susceptibility Mapping	Flood	Ensemble of Machine Learning
Shahri et al. [113]	2021	Disaster Risk Analysis and Management	Disaster Simulations and Modelling	Flood	3D Topographical Models

Table 3. The proposed process-driven and need-oriented analysis of cited literature from preparedness phase.

Author/s	Year	Need and Tasks	Assessments and Activities	Disaster	Technology Used
Chang et al. [35]	2010	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Flood	Conventional Machine Learning
Liao et al. [114]	2010	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Landslide	Geospatial and Satellite Data Modelling
Zhang et al. [115]	2011	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Landslide	Conventional Sensory Equipment
Lin et al. [36]	2013	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Flood	Machine Learning SVM
Devi et al. [116]	2013	Disaster Forecasting	Disaster Pre-Warning System and Prediction	All/Common	Statistical Analysis and Data Mining
Chang et al. [37]	2014	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Flood	Artificial Neural Networks
Dehghani et al. [38]	2014	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Drought	NN and Monte Carlo
Lucieer et al. [57]	2014	Disaster Monitoring	Disaster Resilience Monitoring	Landslide	SfM and UAVs
Asharose et al. [117]	2015	Disaster Awareness	Disaster Workshops and Training	All/Common	Workshops and Educational Tools
Gong et al. [80]	2015	Disaster Awareness	Disaster Workshops and Training	Earthquake	Virtual Reality and Simulations
Moghari and Araghinejad [39]	2015	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Drought	Neural Networks
Hajian et al. [118]	2016	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Wildfire	Monte Carlo Simulation
Asim et al. [40]	2017	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Earthquake	Conventional Machine Learning
Klise et al. [119]	2017	Disaster Monitoring	Disaster Resilience Monitoring	Earthquake	Modelling and Software Package
Hu et al. [81]	2018	Disaster Awareness	Disaster Workshops and Trainings	Flood	Virtual Reality
Seibert et al. [120]	2019	Disaster Monitoring	Water Level Measurements	Flood	Citizen Science
Berkhahn et al. [41]	2019	Disaster Forecasting	Water Level Measurements	Flood	Ensemble Neural Networks
Wang et al. [58]	2019	Disaster Forecasting	Water Level Measurements	Flood	Dilated Casual CNN
De Vitry and Leitao [59]	2020	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Flood	Image Processing and Sensors for Calibration
Mishra et al. [75]	2020	Disaster Forecasting	Disaster Pre-Warning System and Prediction	Flood	Geographic Information System (GIS)
Strobl et al. [121]	2020	Disaster Monitoring	Water Level Measurements	Flood	Citizen Science
Pillai et al. [122]	2021	Disaster Forecasting	Disaster Pre-Warning System and Prediction	All/Common	IoT and Machine Learning
Tamakloe et al. [123]	2021	Disaster Evacuation	Traffic and Routes Management for Effective Evacuation	All/Common	Traffic Network Features based Algorithm

Table 4. The proposed process-driven and need-oriented analysis of cited literature from recovery phase.

Author/s	Year	Need and Tasks	Assessments and Activities	Disaster	Technology Used
Hoshi et al. [69]	2014	Restoration to Normal	Reconstruction Monitoring	Earthquake	Satellite Image Processing and Field Surveys
Baytiyeh [78]	2017	Restoration of Essential Services	Restoring Education and Learning	Earthquake	ICT and Online Learning
Chowdhury et al. [87]	2017	Restoration of Goods Flow	Restoring Logistic Transport Network	All/Common	Drones
Hashemi-Parast et al. [130]	2017	Restoration to Normal	Reconstruction Monitoring	Earthquake	Statistics and Images
Shiraki et al. [131]	2017	Restoration of Transport	Restoring Logistic Transport Network	Earthquake	Resilience Engineering, Heuristic Functions
Yang and Qi [70]	2017	Restoration to Normal	Vegetation Growth Monitoring	Earthquake	Spectral Matching and Cross-Correlogram
Barabadi and Ayele [132]	2018	Recovery Planning	Reconstruction Monitoring	All/Common	Statistical Models and Machine Learning
Contreras et al. [133]	2018	Recovery Planning	Reconstruction Monitoring	Earthquake	Recovery Index based on Spatial Features.
Soulakellis et al. [71]	2020	Recovery Planning	Reconstruction Monitoring	Earthquake	SfM and UAS
Fan et al. [46]	2021	Restoration to Normal	Disaster Recovery Framework	All/Common	Artificial Intelligence and ICT

Table 5. The proposed process-driven and need-oriented analysis of cited literature from response phase.

Author/s	Year	Need and Tasks	Assessments and Activities	Disaster	Technology Used
Poser and Dransch [124]	2010	Disaster Damage Analysis	Rapid Damage Mapping	Flood	VGI
Barrington et al. [125]	2011	Disaster Damage Analysis	Rapid Damage Mapping	Earthquake	Crowdsourced Remote Sensing
Bengtsson et al. [76]	2011	Disaster Information Gathering	Disaster Surveillance and Mapping	Earthquake	Mobile Phone
Choi and Lee [84]	2011	Disaster Monitoring	Disaster Surveillance and Mapping	All/Common	UAVs and Sensors
Aghamohammadi et al. [42]	2013	Disaster Damage Analysis	Disaster Impact and Economic Assessment	Earthquake	Neural Networks
Kao et al. [60]	2013	Disaster Debris Management	Debris Detection and Estimation	Flood	Conventional Image Processing
Lin et al. [36]	2013	Disaster Monitoring	Disaster Surveillance and Mapping	All/Common	Dual-Camera Setup
Houston et al. [47]	2014	Disaster Response	Multiple Assessments	All/Common	Social Media based Framework
Albuquerque et al. [48]	2015	Disaster Information Gathering	Disaster Surveillance and Mapping	Flood	Social Media and VGI
Boccardo et al. [85]	2015	Disaster Information Gathering	Disaster Surveillance and Mapping	All/Common	Unmanned Aerial Vehicles (UAVs)
Jiang and Friedland [126]	2016	Disaster Damage Analysis	Debris Detection and Estimation	Hurricane	Mono-Temporal Image Processing
Kim et al. [82]	2016	Disaster Damage Analysis	Structural Damage Assessment	All/Common	Mobile Virtual Reality
Koyama et al. [61]	2016	Disaster Debris Management	Debris Detection and Estimation	Tsunami	Gradient-Based Optical Flow
Krycasheyeu et al. [49]	2016	Disaster Damage Analysis	Rapid Damage Mapping	Hurricane	Social Media Activity
Bejiga et al. [43]	2017	Disaster Victims Identification	Search and Rescue Operations	Avalanche	CNN, SVM, UAV
Ghosh and Gosavi [127]	2017	Disaster Monitoring	Disaster Impact and Economic Assessment	Earthquake	Semi-Markov Model
Kakooei and Baleghi [62]	2017	Disaster Damage Analysis	Structural Damage Assessment	All/Common	Conventional Image Processing and UAVs
Arnold et al. [86]	2018	Disaster Victims Identification	Search and Rescue Operations	All/Common	Autonomous Aerial Robots and SWARM Intelligence
Galbusera and Giannnopoulos [128]	2018	Disaster Economic Analysis	Disaster Impact and Economic Assessment	All/Common	Input-output Economic models
Vetrivel et al. [63]	2018	Disaster Damage Analysis	Structural Damage Assessment	Earthquake	3D Point Cloud with Deep Learning
Ahmad et al. [44]	2019	Disaster Response	Effective Evacuation Planning	Flood	Deep Learning Models
Aljehani and Inoue [77]	2019	Disaster Scope and Mapping	Disaster Safe Zone and Area Mapping	All/Common	UAVs, and Mobile Phone
Bhola et al. [64]	2019	Disaster Information Gathering	Disaster Safe Zone and Area Mapping	Flood	Computer Vision Approaches
Bird et al. [50]	2019	Disaster Information Gathering	Disaster Communication and Information Sharing	Flood	Social Media
Huang et al. [51]	2019	Disaster Information Gathering	Disaster Communication and Information Sharing	All/Common	Social Media and Deep Learning
Liang [65]	2019	Disaster Damage Analysis	Structural Damage Assessments	All/Common	Deep Learning and Bayesian Optimization
Madichetty and Sridevi [52]	2019	Disaster Damage Analysis	Structural Damage Assessments	All/Common	Social Media Data Mining
Ogie et al. [53]	2019	Disaster Scope and Mapping	Disaster Safe Zone and Area Mapping	Flood	Crowdsourced Social Media
Arif et al. [66]	2020	Disaster Scope and Mapping	Disaster Identification	All/Common	Deep Learning Models
Hao et al. [45]	2020	Disaster Damage Analysis	Structural Damage Assessments	Hurricane	Social Media and Machine Learning
Majumder et al. [129]	2020	Disaster Debris Management	Debris Detection and Estimation	Earthquake	Mathematical Model
Zhai and Peng [67]	2020	Disaster Damage Analysis	Structural Damage Assessments	Hurricane	Google Street View and Deep Learning
Tay et al. [68]	2021	Disaster Damage Analysis	Rapid Damage Mapping	Flood	SAR Images and GIS

	Table 6. Th	ne proposed	disaster	constraints	based a	analysis of	cited	literature	from	prevention	and r	nitigation	phase.
--	-------------	-------------	----------	-------------	---------	-------------	-------	------------	------	------------	-------	------------	--------

Case	Responsiveness	Accuracy	Cost and Hardware Implementation	Generalization	User Friendliness
Bai et al. [24]	×	1	×	x	×
Martyr et al. [97]	×	1	×	×	×
Aronica et al. [98]	×	1	×	×	×
Campos et al. [99]	×	×	×	×	×
Sun et al. [72]	×	×	×	×	1
Cballero and Rahman [100]	×	1	×	×	×
Radianti et al. [101]	×	1	×	1	×
Liu et al. [102]	×	1	×	×	×
Ramakrishnan et al. [103]	×	1	×	1	×
Zhu et al. [83]	×	1	×	×	×
Chen et al. [25]	×	1	×	1	×
Tanaka et al. [104]	NA	NA	×	×	×
Zhang et al. [54]	×	1	×	1	×
Bera et al. [73]	×	1	×	×	×
Komolafe et al. [105]	×	1	×	×	×
Zhang et al. [55]	1	1	×	1	×
Abdi [26]	×	1	×	1	×
Band et al. [27]	×	1	×	1	×
Chowdhuri et al. [28]	×	1	×	1	×
Harirchian and Lahmer [106]	×	1	×	×	×
Lestari et al. [107]	NA	NA	NA	NA	×
Mboga et al. [56]	×	1	×	×	×
Mitsuhara and Shishibori [79]	×	1	×	×	×
Pal et al. [108]	×	1	×	×	×
Pour et al. [109]	NA	NA	×	NA	×
Sansare and Mhaske [74]	×	×	×	×	1
Shahri and Moud [29]	×	1	×	×	×
Ziarh et al. [110]	×	1	×	×	×
Arabameri et al. [30]	×	1	×	1	×
Ndehedehe et al. [31]	×	1	×	×	×
Pal et al. [111]	×	1	×	1	×
Rahman et al. [112]	×	1	×	×	×
Roy et al. [32]	×	1	×	×	×
Saha et al. [33]	×	1	×	1	×
Saha et al. [34]	×	1	×	1	×
Shahri et al. [113]	×	1	X	1	×

sponse phase. From the analysis, it can be observed that "Accuracy" is the most addressed need for the proposed technological solutions; however, all other constraints are least addressed by a significant margin in comparison to "Accuracy". This suggested the lack of collaboration between technology providers and disaster management officials and the lack of proper requirement formulation for the problem to be addressed from a disaster management perspective. A qualitative case study to bring the opinion of disaster management officials into the development process is a potential activity that can be performed in the future.

7. Scope of computer vision in disaster management

A review of technological advancements in disaster management presented in Section 4 has highlighted computer vision as an emerging technology used to facilitate a variety of assessments. From the cited literature, computer vision technology has been used mainly for land-use and land-cover classification [26, 54, 55, 56], water level measurements [58], disaster mapping [64], disaster damage assessments [45, 49, 52, 62, 63, 65, 68, 124, 125], disaster debris management [60] and disaster reconstruction monitoring [69, 71, 130, 132, 133]. However, the technological advancements in edge computing hardware have opened new horizons for computer vision and deep learning technologies towards addressing complex disaster management related problems. The application of computer vision technologies to disaster management related assessments and corresponding potential challenges are discussed under this section.

State-of-the-art CNN models have already proven their utility for land-use and land-cover classification on satellite images with high ac-

curacy; however, on-demand and real-time local mapping is still an unexplored area where UAVs equipped with computer vision algorithms can play their part. From the analysis of literature, machine learning approaches have been abundantly used for disaster hazard zone mapping; however, the scope of computer vision technologies on satellite and airborne images has not been explored to its potential. Monitoring disaster prevention structures such as Dams is another domain where computer vision can be deployed for real-time damage detection and water-level measurements. At the preparedness phase, disaster monitoring and early warning systems can make use of computer vision based approaches for disaster detection. Real-time water level measurement at disaster vulnerable points using computer vision is one typical example where computer vision has already been used to issue early warnings for flood disasters. However, most of the studies were performed on a local scale and using conventional image processing techniques. For disaster forecasting, incorporation of visual data and respective computer vision analysis into existing time-series numerical data based models is one potential domain that can be explored.

At the response phase, on-demand quick mapping of disaster to identify the scope and extent of disaster using a UAV equipped with computer vision algorithms is a potential domain to explore. As of now, disaster mapping is done either by manned helicopters, ground communications, or satellite images. However, UAVs and computer vision have the potential to make this process cost economical, quick and more accurate to facilitate post-disaster response activities. In a similar scope, disaster victims identification is another assessment that can be facilitated by UAVs and computer vision algorithms (i.e., object detection, object segmentation). Post-disaster structural damage estimation is the assessment where computer vision has been deployed the most using

Table 7.	. The	proposed	disaster	constraints	based	analysi	s of	cited	literature	from	prepa	aredness	phase
----------	-------	----------	----------	-------------	-------	---------	------	-------	------------	------	-------	----------	-------

Case	Responsiveness	Accuracy	Cost and Hardware	Generalization	User Friendliness
Chang et al [25]	~	1		1	~
Lizo et al [114]	~	· ·	*	v	~
There et al [115]	~	•	~	~	~
Zhang et al. [115]	x	<i>,</i>	V	×	X
Lin et al. [36]	×	1	X	✓	X
Devi et al. [116]	X	×	×	1	1
Chang et al. [37]	×	1	×	×	×
Dehghani et al. [38]	×	1	х	1	х
Lucieer et al. [57]	×	1	x	×	×
Asharose et al. [117]	NA	NA	NA	NA	1
Gong et al. [80]	×	×	×	1	1
Moghari and Araghinejad [39]	×	1	×	1	×
Hajian et al. [118]	×	1	×	×	×
Asim et al. [40]	×	1	×	×	×
Klise et al. [119]	×	1	×	1	1
Hu et al. [81]	×	1	×	×	1
Seibert et al. [120]	×	1	×	×	1
Berkhahn et al. [41]	1	1	×	×	×
Wang et al. [58]	×	1	×	1	×
De Vitry and Leitao [59]	×	1	×	×	×
Mishra et al. [75]	×	×	×	×	1
Strobl et al. [121]	×	1	x	×	1
Pillai et al. [122]	×	1	1	×	1
Tamakloe et al. [123]	1	1	×	×	×

Table 8. The proposed disaster constraints based analysis of cited literature from recovery phase.

Case	Responsiveness	Accuracy	Cost and Hardware Implementation	Generalization	User Friendliness
Hoshi et al. [69]	×	1	×	×	×
Baytiyeh [78]	NA	NA	NA	NA	NA
Chowdhury et al. [87]	×	1	1	×	×
Hashemi-Parast et al. [130]	×	1	×	×	×
Shiraki et al. [131]	×	×	×	×	×
Yang and Qi [70]	×	1	×	1	×
Barabadi and Ayele [132]	×	1	×	×	×
Contreras et al. [133]	×	1	×	×	×
Soulakellis et al. [71]	×	1	×	×	×
Fan et al. [46]	NA	NA	×	NA	×

satellite images and UAV captured images. State-of-the-art technologies such as SfM and 3D reconstruction based on point clouds have already been efficiently used. However, detailed structural damage assessments and building re-usability assessments at a local scale are least explored from a computer vision perspective and can be benefited. Disaster debris detection and estimation is an important assessment where computer vision technologies are least applied. For example, assessment of blockage at culverts using computer vision algorithms to avoid floods is one potential domain [142]. At the recovery phase, reconstruction monitoring from satellite images is the most explored assessment where computer vision is deployed. However, on-demand reconstruction monitoring at the local scale is still unexplored. Vegetation growth monitoring is another assessment where computer vision can be used in the same scope as reconstruction monitoring.

One of the highlighted challenges observed from cited literature is the availability of comprehensive benchmark visual datasets to address the disaster management related assessments from a computer vision perspective. It was observed that for the assessments where benchmark datasets were available, state-of-the-art computer vision algorithms are already deployed (e.g., land-use classification, structural damage mapping). Otherwise, most of the studies were performed using custom datasets with limited scope. In addition, dealing with diverse weather conditions, variable lighting conditions, and camera viewpoint problems are some highlighted challenges in deploying computer vision based solutions for addressing disaster management assessments.

8. Conclusion

A review of benchmark studies was presented using a proposed process-driven and need-oriented framework to highlight the technological trends in disaster management. The proposed framework helped in better classification of literature and finding potential research gaps. Machine learning, UAVs, AI, ICT, robotics, Big Data, virtual reality, augmented reality, social media and computer vision were reported as the highlighted technologies deployed to address disaster management assessments. Disaster hazard zone mapping, land-use classification, disaster forecasting, water level measurement, disaster damage assessments, and reconstruction monitoring were the reported assessments where most technological solutions were proposed. A list of disaster related constraints was formulated to align the solutions against the disaster management requirements. Accuracy was reported as the most fulfilled disaster requirement in literature, while responsiveness, generalization, cost and hardware implementation, and user-friendliness were rarely discussed. Proper requirement formulation related to addressed disaster management assessment for effective outcome was found consistently lacking. Overall, the availability of comprehensive benchmark datasets was found missing and most of the studies were performed for a local utility. Finally, the scope of computer vision in facilitating different disaster management related assessments was explored and potential challenges were highlighted as future research directions. Land-use and land-cover classification, water level measurements, disaster mapping, disaster damage assessments, disaster debris management and disas-

Table 9. The proposed disaster constraints based analysis of cited literature from response phase.

Case	Responsiveness	Accuracy	Cost and Hardware Implementation	Generalization	User Friendliness
Poser and Dransch [124]	×	1	х	×	×
Barrington et al. [125]	NA	NA	NA	NA	1
Bengtsson et al. [76]	×	1	×	1	×
Choi and Lee [84]	×	×	1	×	1
Aghamohammadi et al. [42]	×	1	×	×	×
Kao et al. [60]	1	1	×	×	×
Lin et al. [36]	×	×	×	×	×
Houston et al. [47]	NA	NA	NA	NA	1
Albuquerque et al. [48]	×	1	×	×	×
Boccardo et al. [85]	×	1	1	×	1
Jiang and Friedland [126]	1	1	×	×	×
Kim et al. [82]	×	1	×	1	1
Koyama et al. [61]	×	1	×	×	×
Krycasheyeu et al. [49]	×	1	×	×	×
Bejiga et al. [43]	1	1	×	1	×
Ghosh and Gosavi [127]	×	1	×	×	×
Kakooei and Baleghi [62]	×	1	×	×	×
Arnold et al. [86]	×	1	×	1	×
Galbusera and Giannnopoulos [128]	NA	NA	NA	NA	×
Vetrivel et al. [63]	1	1	×	1	×
Ahmad et al. [44]	×	1	×	1	×
Aljehani and Inoue [77]	×	1	1	1	1
Bhola et al. [64]	×	1	×	1	×
Bird et al. [50]	NA	NA	NA	NA	1
Huang et al. [51]	×	1	×	1	×
Liang [65]	×	1	×	×	×
Madichetty and Sridevi [52]	×	1	×	1	×
Ogie et al. [53]	×	1	×	1	×
Arif et al. [66]	×	1	×	1	×
Hao et al. [45]	×	1	×	×	×
Majumder et al. [129]	×	×	×	×	1
Zhai and Peng [67]	×	1	×	×	×
Tay et al. [68]	×	×	×	×	×

ter reconstruction monitoring were reported as highlighted assessments where computer vision can potentially be used in the future.

Declarations

Author contribution statement

All authors listed have significantly contributed to the development and the writing of this article.

Funding statement

This work was supported by Higher Education Commission (HEC) of Pakistan and the University of Wollongong (UOW).

Data availability statement

No data was used for the research described in the article.

Declaration of interests statement

The authors declare no conflict of interest.

Additional information

No additional information is available for this paper.

Acknowledgements

I would like to thank Associate Professor Wanqing Li for his contribution to the early conceptualization of the proposed framework.

References

- T.V. Kumar, K. Sud, AI and Robotics in Disaster Studies, 1st ed., Palgrave Macmillan, Singapore, 2020.
- [2] P. Mohan, H. Mittal, Review of ICT usage in disaster management, Int. J. Inf. Technol. 12 (2020) 955–962.
- [3] E. Quarantelli, What is a disaster: perspectives on the question, Disaster Prev. Manag.: Int. J. 8 (5) (1999) 370–452.
- [4] P. Blaikie, T. Cannon, I. Davis, B. Wisner, At Risk: Natural Hazards, People's Vulnerability and Disasters, 2nd ed., Routledge, 2005.
- [5] A. Dewan, Floods in a Megacity: Geospatial Techniques in Assessing Hazards, Risk and Vulnerability, 1st ed., Springer, Netherlands, 2013.
- [6] L. Bull-Kamanga, K. Diagne, A. Lavell, E. Leon, F. Lerise, H. MacGregor, A. Maskrey, M. Meshack, M. Pelling, H. Reid, et al., From everyday hazards to disasters: the accumulation of risk in urban areas, Environ. Urban. 15 (1) (2003) 193–204.
- [7] A.A. Shahri, F.M. Moud, Liquefaction potential analysis using hybrid multiobjective intelligence model, Environ. Earth Sci. 79 (19) (2020) 1–17.
- [8] K. Eshghi, R.C. Larson, Disasters: lessons from the past 105 years, Disaster Prev. Manag.: Int. J. 17 (1) (2008) 62–82.
- [9] J.T. Rodríguez, B. Vitoriano, D. Gómez, J. Montero, Classification of disasters and emergencies under bipolar knowledge representation, in: Decision Aid Models for Disaster Management and Emergencies, Springer, 2013, pp. 209–232.
- [10] A. Nofal, I. AlFayyad, N. AlJerian, J. Alowais, M. AlMarshady, A. Khan, H. Heena, A.S. AlSarheed, A. Abu-Shaheen, Knowledge and preparedness of healthcare providers towards bioterrorism, BMC Health Serv. Res. 21 (1) (2021) 1–13.
- [11] D. Guha-Sapir, Em-Dat: The Emergency Events Database, Universite catholique de Louvain (UCL)—CRED, 2021.
- [12] D. Guha-Sapir, F. Vos, R. Below, S. Ponserre, Annual disaster statistical review 2011: the numbers and trends, Tech. Rep., Centre for Research on the Epidemiology of Disasters (CRED), 2012.
- [13] D.P. Coppola, Introduction to International Disaster Management, 3rd ed., Elsevier, 2015.
- [14] F. Kraas, Megacities as global risk areas, in: Urban Ecology: an International Perspective on the Interaction Between Humans and Nature, Springer, New York, 2008, pp. 583–596.
- [15] S. Akter, S.F. Wamba, Big data and disaster management: a systematic review and agenda for future research, Ann. Oper. Res. 283 (1) (2019) 939–959.

- [16] W.N. Carter, Disaster Management: A Disaster Manager's Handbook, Asian Development Bank, 6 ADB Avenue, Mandaluyong City 1550 Metro Manila, Philippines, 2008.
- [17] B. Wisner, P. Blaikie, T. Cannon, I. Davis, At Risk: Natural Hazards, People's Vulnerability and Disasters, 2nd ed., Routledge, Abingdon, United Kingdom, 2004.
- [18] S.L. Cutter, Vulnerability to environmental hazards, Prog. Hum. Geogr. 20 (4) (1996) 529–539.
- [19] K. Hewitt, Regions of Risk: A Geographical Introduction to Disasters, 1st ed., Longman, Essex, United Kingdom, 1997.
- [20] S. Greiving, M. Fleischhauer, J. Lückenkötter, A methodology for an integrated risk assessment of spatially relevant hazards, J. Environ. Plan. Manag. 49 (1) (2006) 1–19.
- [21] B. Affeltranger, Living with Risk: A Global Review of Disaster Reduction Initiatives, ISDR, 2004.
- [22] D. Seaberg, L. Devine, J. Zhuang, A review of game theory applications in natural disaster management research, Nat. Hazards 89 (3) (2017) 1461–1483.
- [23] D. Mishra, S. Kumar, E. Hassini, Current trends in disaster management simulation modelling research, Ann. Oper. Res. 283 (1) (2019) 1387–1411.
- [24] S.-B. Bai, J. Wang, G.-N. Lü, P.-G. Zhou, S.-S. Hou, S.-N. Xu, GIS-based logistic regression for landslide susceptibility mapping of the Zhongxian segment in the three gorges area, China, Geomorphology 115 (1–2) (2010) 23–31.
- [25] W. Chen, J. Peng, H. Hong, H. Shahabi, B. Pradhan, J. Liu, A.-X. Zhu, X. Pei, Z. Duan, Landslide susceptibility modelling using GIS-based machine learning techniques for Chongren County, Jiangxi Province, China, Sci. Total Environ. 626 (2018) 1121–1135.
- [26] A.M. Abdi, Land cover and land use classification performance of machine learning algorithms in a boreal landscape using sentinel-2 data, GISci. Remote Sens. 57 (1) (2020) 1–20.
- [27] S.S. Band, S. Janizadeh, S. Chandra Pal, A. Saha, R. Chakrabortty, A.M. Melesse, A. Mosavi, Flash flood susceptibility modeling using new approaches of hybrid and ensemble tree-based machine learning algorithms, Remote Sens. 12 (21) (2020) 3568.
- [28] I. Chowdhuri, S.C. Pal, A. Arabameri, A. Saha, R. Chakrabortty, T. Blaschke, B. Pradhan, S. Band, et al., Implementation of artificial intelligence based ensemble models for gully erosion susceptibility assessment, Remote Sens. 12 (21) (2020) 3620.
- [29] A.A. Shahri, F.M. Moud, Landslide susceptibility mapping using hybridized block modular intelligence model, Bull. Eng. Geol. Environ. 80 (1) (2021) 267–284.
- [30] A. Arabameri, S. Chandra Pal, R. Costache, A. Saha, F. Rezaie, A. Seyed Danesh, B. Pradhan, S. Lee, N.-D. Hoang, Prediction of gully erosion susceptibility mapping using novel ensemble machine learning algorithms, Geomat. Nat. Hazards Risk 12 (1) (2021) 469–498.
- [31] C.E. Ndehedehe, V.G. Ferreira, N.O. Agutu, A.O. Onojeghuo, O. Okwuashi, H.T. Kassahun, A. Dewan, What if the rains do not come?, J. Hydrol. 595 (2021) 126040.
- [32] P. Roy, S.C. Pal, A. Arabameri, F. Rezaie, R. Chakrabortty, I. Chowdhuri, A. Saha, S. Malik, B. Das, Climate and land use change induced future flood susceptibility assessment in a sub-tropical region of India, Soft Comput. 25 (8) (2021) 5925–5949.
- [33] A. Saha, S.C. Pal, A. Arabameri, I. Chowdhuri, F. Rezaie, R. Chakrabortty, P. Roy, M. Shit, Optimization modelling to establish false measures implemented with exsitu plant species to control gully erosion in a monsoon-dominated region with novel in-situ measurements, J. Environ. Manag. 287 (2021) 112284.
- [34] A. Saha, S.C. Pal, A. Arabameri, T. Blaschke, S. Panahi, I. Chowdhuri, R. Chakrabortty, R. Costache, A. Arora, Flood susceptibility assessment using novel ensemble of hyperpipes and support vector regression algorithms, Water 13 (2) (2021) 241.
- [35] L.-C. Chang, H.-Y. Shen, Y.-F. Wang, J.-Y. Huang, Y.-T. Lin, Clustering-based hybrid inundation model for forecasting flood inundation depths, J. Hydrol. 385 (1–4) (2010) 257–268.
- [36] C.-W. Lin, Y.-P. Hung, W.-K. Hsu, W.-L. Chiang, C.-W. Chen, The construction of a high-resolution visual monitoring for hazard analysis, Nat. Hazards 65 (3) (2013) 1285–1292.
- [37] F.-J. Chang, P.-A. Chen, Y.-R. Lu, E. Huang, K.-Y. Chang, Real-time multi-stepahead water level forecasting by recurrent neural networks for urban flood control, J. Hydrol. 517 (2014) 836–846.
- [38] M. Dehghani, B. Saghafian, F. Nasiri Saleh, A. Farokhnia, R. Noori, Uncertainty analysis of streamflow drought forecast using artificial neural networks and Monte-Carlo simulation, Int. J. Climatol. 34 (4) (2014) 1169–1180.
- [39] S.M. Hosseini-Moghari, S. Araghinejad, Monthly and seasonal drought forecasting using statistical neural networks, Environ. Earth Sci. 74 (1) (2015) 397–412.
- [40] K. Asim, F. Martínez-Álvarez, A. Basit, T. Iqbal, Earthquake magnitude prediction in Hindukush region using machine learning techniques, Nat. Hazards 85 (1) (2017) 471–486.
- [41] S. Berkhahn, L. Fuchs, I. Neuweiler, An ensemble neural network model for realtime prediction of urban floods, J. Hydrol. 575 (2019) 743–754.
- [42] H. Aghamohammadi, M. Mesgari, A. Mansourian, D. Molaei, Seismic human loss estimation for an earthquake disaster using neural network, Int. J. Environ. Sci. Technol. 10 (5) (2013) 931–939.
- [43] M.B. Bejiga, A. Zeggada, A. Nouffidj, F. Melgani, A convolutional neural network approach for assisting avalanche search and rescue operations with UAV imagery, Remote Sens. 9 (2) (2017) 100.

- [44] K. Ahmad, K. Pogorelov, M. Riegler, O. Ostroukhova, P. Halvorsen, N. Conci, R. Dahyot, Automatic detection of passable roads after floods in remote sensed and social media data, Signal Process. Image Commun. 74 (2019) 110–118.
- [45] H. Hao, Y. Wang, Leveraging multimodal social media data for rapid disaster damage assessment, Int. J. Disaster Risk Reduct. (2020) 101760.
- [46] C. Fan, C. Zhang, A. Yahja, A. Mostafavi, Disaster city digital twin: a vision for integrating artificial and human intelligence for disaster management, Int. J. Inf. Manag. (2021) 102049.
- [47] J.B. Houston, J. Hawthorne, M.F. Perreault, E.H. Park, M. Goldstein Hode, M.R. Halliwell, S.E. Turner McGowen, R. Davis, S. Vaid, J.A. McElderry, et al., Social media and disasters: a functional framework for social media use in disaster planning, response, and research, Disasters 39 (1) (2015) 1–22.
- [48] J.P. De Albuquerque, B. Herfort, A. Brenning, A. Zipf, A geographic approach for combining social media and authoritative data towards identifying useful information for disaster management, Int. J. Geogr. Inf. Sci. 29 (4) (2015) 667–689.
- [49] Y. Kryvasheyeu, H. Chen, N. Obradovich, E. Moro, P. Van Hentenryck, J. Fowler, M. Cebrian, Rapid assessment of disaster damage using social media activity, Sci. Adv. 2 (3) (2016) e1500779.
- [50] D. Bird, M. Ling, K. Haynes, Flooding Facebook-the use of social media during the Queensland and Victorian floods, Aust. J. Emerg. Manag. 27 (1) (2019) 27–33.
- [51] X. Huang, Z. Li, C. Wang, H. Ning, Identifying disaster related social media for rapid response: a visual-textual fused CNN architecture, Int. J. Dig. Earth (2019) 1–23.
- [52] S. Madichetty, M. Sridevi, Disaster damage assessment from the tweets using the combination of statistical features and informative words, Soc. Netw. Anal. Min. 9 (1) (2019) 42.
- [53] R.I. Ogie, R.J. Clarke, H. Forehead, P. Perez, Crowdsourced social media data for disaster management: lessons from the PetaJakarta.org project, Comput. Environ. Urban Syst. 73 (2019) 108–117.
- [54] C. Zhang, I. Sargent, X. Pan, H. Li, A. Gardiner, J. Hare, P.M. Atkinson, An objectbased convolutional neural network (OCNN) for urban land use classification, Remote Sens. Environ. 216 (2018) 57–70.
- [55] C. Zhang, I. Sargent, X. Pan, H. Li, A. Gardiner, J. Hare, P.M. Atkinson, Joint deep learning for land cover and land use classification, Remote Sens. Environ. 221 (2019) 173–187.
- [56] N. Mboga, T. Grippa, S. Georganos, S. Vanhuysse, B. Smets, O. Dewitte, E. Wolff, M. Lennert, Fully convolutional networks for land cover classification from historical panchromatic aerial photographs, ISPRS J. Photogramm. Remote Sens. 167 (2020) 385–395.
- [57] A. Lucieer, S.M.d. Jong, D. Turner, Mapping landslide displacements using structure from motion (SfM) and image correlation of multi-temporal UAV photography, Prog. Phys. Geogr. 38 (1) (2014) 97–116.
- [58] J.-H. Wang, G.-F. Lin, M.-J. Chang, I.-H. Huang, Y.-R. Chen, Real-time waterlevel forecasting using dilated causal convolutional neural networks, Water Resour. Manag. 33 (11) (2019) 3759–3780.
- [59] M.M. de Vitry, J.P. Leitão, The potential of proxy water level measurements for calibrating urban pluvial flood models, Water Res. (2020) 115669.
- [60] H.-M. Kao, H. Ren, C.-S. Lee, Y.-L. Chen, Y.-S. Lin, Y. Su, Monitoring debris flows using spatial filtering and entropy determination approaches, Terr. Atmos. Ocean. Sci. 24 (5) (2013).
- [61] C.N. Koyama, H. Gokon, M. Jimbo, S. Koshimura, M. Sato, Disaster debris estimation using high-resolution polarimetric stereo-SAR, ISPRS J. Photogramm. Remote Sens. 120 (2016) 84–98.
- [62] M. Kakooei, Y. Baleghi, Fusion of satellite, aircraft, and UAV data for automatic disaster damage assessment, Int. J. Remote Sens. 38 (8–10) (2017) 2511–2534.
- [63] A. Vetrivel, M. Gerke, N. Kerle, F. Nex, G. Vosselman, Disaster damage detection through synergistic use of deep learning and 3d point cloud features derived from very high resolution oblique aerial images, and multiple-kernel-learning, ISPRS J. Photogramm. Remote Sens. 140 (2018) 45–59.
- [64] P.K. Bhola, B.B. Nair, J. Leandro, S.N. Rao, M. Disse, Flood inundation forecasts using validation data generated with the assistance of computer vision, J. Hydroinform. 21 (2) (2019) 240–256.
- [65] X. Liang, Image-based post-disaster inspection of reinforced concrete bridge systems using deep learning with Bayesian optimization, Comput.-Aided Civ. Infrastruct. Eng. 34 (5) (2019) 415–430.
- [66] Arif, M.A. Amin, A.A. Ali, A.M. Rahman, et al., Visual attention-based comparative study on disaster detection from social media images, Innov. Syst. Softw. Eng. (2020) 1–11.
- [67] W. Zhai, Z.-R. Peng, Damage assessment using Google street view: evidence from hurricane Michael in Mexico beach, Florida, Appl. Geogr. 123 (2020) 102252.
- [68] C.W. Tay, S.-H. Yun, S.T. Chin, A. Bhardwaj, J. Jung, E.M. Hill, Rapid flood and damage mapping using synthetic aperture radar in response to typhoon Hagibis, Japan, Sci. Data 7 (1) (2020) 1–9.
- [69] T. Hoshi, O. Murao, K. Yoshino, F. Yamazaki, M. Estrada, Post-disaster urban recovery monitoring in pisco after the 2007 Peru earthquake using satellite image, J. Disaster Res. 9 (6) (2014) 1059–1068.
- [70] W. Yang, W. Qi, Spatial-temporal dynamic monitoring of vegetation recovery after the Wenchuan earthquake, IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens. 10 (3) (2017) 868–876.

- [71] N. Soulakellis, C. Vasilakos, S. Chatzistamatis, D. Kavroudakis, G. Tataris, E.-E. Papadopoulou, A. Papakonstantinou, O. Roussou, T. Kontos, Post-earthquake recovery phase monitoring and mapping based on UAS data, ISPRS Int.l J. Geo-Inf. 9 (7) (2020) 447.
- [72] D. Sun, D. Zhang, X. Cheng, Framework of national non-structural measures for flash flood disaster prevention in China, Water 4 (1) (2012) 272–282.
- [73] A. Bera, B.P. Mukhopadhyay, D. Das, Landslide hazard zonation mapping using multi-criteria analysis with the help of GIS techniques: a case study from eastern Himalayas, Namchi, South Sikkim, Nat. Hazards 96 (2) (2019) 935–959.
- [74] D.A. Sansare, S.Y. Mhaske, Natural hazard assessment and mapping using remote sensing and QGIS tools for Mumbai city, India, Nat. Hazards 100 (3) (2020) 1117–1136.
- [75] S. Mishra, S. Chander, R. Pradhan, A.K. Dubey, M.P. Oza, S.A. Sharma, Webgis for water level monitoring and flood forecasting using open source technology, J. Geomat. 14 (1) (2020).
- [76] L. Bengtsson, X. Lu, A. Thorson, R. Garfield, J. Von Schreeb, Improved response to disasters and outbreaks by tracking population movements with mobile phone network data: a post-earthquake geospatial study in Haiti, PLoS Med. 8 (8) (2011) e1001083.
- [77] M. Aljehani, M. Inoue, Safe map generation after a disaster, assisted by an unmanned aerial vehicle tracking system, IEEJ Trans. Electr. Electron. Eng. 14 (2) (2019) 271–282.
- [78] H. Baytiyeh, Online learning during post-earthquake school closures, Disaster Prev. Manag.: Int. J. 27 (2) (2018) 215–227.
- [79] H. Mitsuhara, M. Shishibori, Comparative experiments on simulated tornado experience via virtual reality and augmented reality, J. Inf. Syst. Educ. 19 (1) (2020) 21–31.
- [80] X. Gong, Y. Liu, Y. Jiao, B. Wang, J. Zhou, H. Yu, A novel earthquake education system based on virtual reality, IEICE Trans. Inf. Syst. 98 (12) (2015) 2242–2249.
- [81] Y. Hu, J. Zhu, W. Li, Y. Zhang, Q. Zhu, H. Qi, H. Zhang, Z. Cao, W. Yang, P. Zhang, Construction and optimization of three-dimensional disaster scenes within mobile virtual reality, ISPRS Int.I J. Geo-Inf. 7 (6) (2018) 215.
- [82] W. Kim, N. Kerle, M. Gerke, Mobile augmented reality in support of building damage and safety assessment, Nat. Hazards Earth Syst. Sci. 16 (1) (2016) 287.
- [83] Z.J. Zhu, A.Z. Jiang, J. Lai, Y. Xiang, B. Baird, E. McBean, Towards efficient use of an unmanned aerial vehicle for urban flood monitoring, J. Water Manag. Model. (2017) 1–7.
- [84] K. Choi, I. Lee, A UAV-based close-range rapid aerial monitoring system for emergency responses, Int. Arch. Photogramm. Remote Sens. Spat. Inf. Sci. 38 (2011) 247–252.
- [85] P. Boccardo, F. Chiabrando, F. Dutto, F.G. Tonolo, A. Lingua, UAV deployment exercise for mapping purposes: evaluation of emergency response applications, Sensors 15 (7) (2015) 15717–15737.
- [86] R.D. Arnold, H. Yamaguchi, T. Tanaka, Search and rescue with autonomous flying robots through behavior-based cooperative intelligence, J. Int. Humanit. Action 3 (1) (2018) 1–18.
- [87] S. Chowdhury, A. Emelogu, M. Marufuzzaman, S.G. Nurre, L. Bian, Drones for disaster response and relief operations: a continuous approximation model, Int. J. Prod. Econ. 188 (2017) 167–184.
- [88] W. Sun, P. Bocchini, B.D. Davison, Applications of artificial intelligence for disaster management, Nat. Hazards (2020) 1–59.
- [89] M. Yu, C. Yang, Y. Li, Big data in natural disaster management: a review, Geosciences 8 (5) (2018) 165.
- [90] I.H. Sawalha, A contemporary perspective on the disaster management cycle, Foresight 22 (4) (2020) 469–482.
- [91] H. Khan, L.G. Vasilescu, A. Khan, et al., Disaster management cycle-a theoretical approach, J. Manag. Mark. 6 (1) (2008) 43–50.
- [92] L. Lopez-Fuentes, J. van de Weijer, M. González-Hidalgo, H. Skinnemoen, A.D. Bagdanov, Review on computer vision techniques in emergency situations, Multimed. Tools Appl. 77 (13) (2018) 17069–17107.
- [93] R. Haigh, Disaster Management Lifecycle, University of Salford, England, 2017.
- [94] H. Khan, A. Khan, Natural hazards and disaster management in Pakistan, 2008.
- [95] M.D. McInnes, D. Moher, B.D. Thombs, T.A. McGrath, P.M. Bossuyt, T. Clifford, J.F. Cohen, J.J. Deeks, C. Gatsonis, L. Hooft, Preferred reporting items for a systematic review and meta-analysis of diagnostic test accuracy studies: the PRISMA-DTA statement, JAMA 319 (4) (2018) 388–396.
- [96] G. Galindo, R. Batta, Review of recent developments in or/ms research in disaster operations management, Eur. J. Oper. Res. 230 (2) (2013) 201–211.
- [97] R. Martyr, J. Dietrich, J.J. Westerink, P. Kerr, C. Dawson, J. Smith, H. Pourtaheri, N. Powell, M. Van Ledden, S. Tanaka, et al., Simulating hurricane storm surge in the lower Mississippi River under varying flow conditions, J. Hydraul. Eng. 139 (5) (2013) 492–501.
- [98] G. Aronica, F. Franza, P. Bates, J. Neal, Probabilistic evaluation of flood hazard in urban areas using Monte Carlo simulation, Hydrol. Process. 26 (26) (2012) 3962–3972.
- [99] V. Campos, R. Bandeira, A. Bandeira, A method for evacuation route planning in disaster situations, Proc., Soc. Behav. Sci. 54 (2012) 503–512.
- [100] W.L. Caballero, A. Rahman, Application of Monte Carlo simulation technique for flood estimation for two catchments in New South Wales, Australia, Nat. Hazards 74 (3) (2014) 1475–1488.

- [101] J. Radianti, O.-C. Granmo, P. Sarshar, M. Goodwin, J. Dugdale, J.J. Gonzalez, A spatio-temporal probabilistic model of hazard-and crowd dynamics for evacuation planning in disasters, Appl. Intell. 42 (1) (2015) 3–23.
- [102] Q. Liu, Y. Qin, Y. Zhang, Z. Li, A coupled 1d–2d hydrodynamic model for flood simulation in flood detention basin, Nat. Hazards 75 (2) (2015) 1303–1325.
- [103] S. Ramakrishnan, X. Wang, J. Sanjayan, J. Wilson, Thermal performance of buildings integrated with phase change materials to reduce heat stress risks during extreme heatwave events, Appl. Energy 194 (2016) 410–421.
- [104] H. Tanaka, D. Sasaki, Y. Ono, Proposed requirement definition method for developing global disaster database based on various means of data collection, J. Disaster Res. 13 (6) (2018) 1015–1023.
- [105] A.A. Komolafe, S. Herath, R. Avtar, Establishment of detailed loss functions for the urban flood risk assessment in Chao Phraya river basin, Thailand, Geomat. Nat. Hazards Risk 10 (1) (2019) 633–650.
- [106] E. Harirchian, T. Lahmer, Developing a hierarchical type-2 fuzzy logic model to improve rapid evaluation of earthquake hazard safety of existing buildings, Structures 28 (2020) 1384–1399.
- [107] P. Lestari, R. Ritonga, P. Ruliana, C.C.B. Barus, Disaster communication uses field training exercise simulation as an important aspect of disaster risk reduction, J. Komun.: Malays. J. Commun. 36 (1) (2020).
- [108] S.C. Pal, A. Arabameri, T. Blaschke, I. Chowdhuri, A. Saha, R. Chakrabortty, S. Lee, S. Band, et al., Ensemble of machine-learning methods for predicting gully erosion susceptibility, Remote Sens. 12 (22) (2020) 3675.
- [109] S.H. Pour, A.K. Abd Wahab, S. Shahid, M. Asaduzzaman, A. Dewan, Low impact development techniques to mitigate the impacts of climate-change-induced urban floods: current trends, issues and challenges, Sustain. Cities Soc. 62 (2020) 102373.
- [110] G.F. Ziarh, M. Asaduzzaman, A. Dewan, M.S. Nashwan, S. Shahid, Integration of catastrophe and entropy theories for flood risk mapping in Peninsular Malaysia, J. Flood Risk Manag. 14 (1) (2021) e12686.
- [111] S.C. Pal, R. Chakrabortty, P. Roy, I. Chowdhuri, B. Das, A. Saha, M. Shit, Changing climate and land use of 21st century influences soil erosion in India, Gondwana Res. 94 (2021) 164–185.
- [112] M. Rahman, C. Ningsheng, M.M. Islam, G.I. Mahmud, H.R. Pourghasemi, M. Alam, M.A. Rahim, M.A. Baig, A. Bhattacharjee, A. Dewan, Development of flood hazard map and emergency relief operation system using hydrodynamic modeling and machine learning algorithm, J. Clean. Prod. (2021) 127594.
- [113] A. Abbaszadeh Shahri, A. Kheiri, A. Hamzeh, Subsurface topographic modeling using geospatial and data driven algorithm, ISPRS Int.l J. Geo-Inf. 10 (5) (2021) 341.
- [114] Z. Liao, Y. Hong, J. Wang, H. Fukuoka, K. Sassa, D. Karnawati, F. Fathani, Prototyping an experimental early warning system for rainfall-induced landslides in Indonesia using satellite remote sensing and geospatial datasets, Landslides 7 (3) (2010) 317–324.
- [115] Y. Zhang, H. Li, Q. Sheng, K. Wu, G. Chen, Real time remote monitoring and prewarning system for highway landslide in mountain area, J. Environ. Sci. 23 (2011) S100–S105.
- [116] B.R. Devi, K. Rao, S. Setty, M. Rao, Disaster prediction system using IBM SPSS data mining tool, Int. J. Eng. Trends Technol. (IJETT) 4 (2013) 3352–3357.
- [117] Asharose, I. Saizen, P.K.C. Sasi, et al., Awareness workshop as an effective tool and approach for education in disaster risk reduction: a case study from Tamil Nadu, India, Sustainability 7 (7) (2015) 8965–8984.
- [118] M. Hajian, E. Melachrinoudis, P. Kubat, Modeling wildfire propagation with the stochastic shortest path: a fast simulation approach, Environ. Model. Softw. 82 (2016) 73–88.
- [119] K.A. Klise, M. Bynum, D. Moriarty, R. Murray, A software framework for assessing the resilience of drinking water systems to disasters with an example earthquake case study, Environ. Model. Softw. 95 (2017) 420–431.
- [120] J. Seibert, B. Strobl, S. Etter, P. Hummer, H. van Meerveld, Virtual staff gauges for crowd-based stream level observations, Front. Earth Sci. 7 (2019) 70.
- [121] B. Strobl, S. Etter, I. van Meerveld, J. Seibert, Accuracy of crowdsourced streamflow and stream level class estimates, Hydrol. Sci. J. 65 (5) (2020) 823–841.
- [122] A.S. Pillai, G.S. Chandraprasad, A.S. Khwaja, A. Anpalagan, A service oriented IoT architecture for disaster preparedness and forecasting system, Int. Things (2021) 100076.
- [123] R. Tamakloe, J. Hong, J. Tak, D. Park, Finding evacuation routes using traffic and network structure information, Transp. Res., Part D, Transp. Environ. 95 (2021) 102853.
- [124] K. Poser, D. Dransch, Volunteered geographic information for disaster management with application to rapid flood damage estimation, Geomatica 64 (1) (2010) 89–98.
- [125] L. Barrington, S. Ghosh, M. Greene, S. Har-Noy, J. Berger, S. Gill, A.Y.-M. Lin, C. Huyck, Crowdsourcing earthquake damage assessment using remote sensing imagery, Ann. Geophys. 54 (6) (2011).
- [126] S. Jiang, C.J. Friedland, Automatic urban debris zone extraction from posthurricane very high-resolution satellite and aerial imagery, Geomat. Nat. Hazards Risk 7 (3) (2016) 933–952.
- [127] S. Ghosh, A. Gosavi, A semi-Markov model for post-earthquake emergency response in a smart city, Control Theory Technol. 15 (1) (2017) 13–25.
- [128] L. Galbusera, G. Giannopoulos, On input-output economic models in disaster impact assessment, Int. J. Disaster Risk Reduct. 30 (2018) 186–198.

- [129] A. Majumder, K. Pradhan, S.S. Danewalia, B.K. Sett, A review and modelling on the critical management of the disaster debris of earthquake in Bhutan, Reg. Sci. Policy Pract. 12 (3) (2020) 477–491.
- [130] S.O. Hashemi-Parast, F. Yamazaki, W. Liu, Monitoring and evaluation of the urban reconstruction process in Bam, Iran, after the 2003 m w 6.6 earthquake, Nat. Hazards 85 (1) (2017) 197–213.
- [131] W. Shiraki, K. Takahashi, H. Inomo, C. Isouchi, A proposed restoration strategy for road networks after an earthquake disaster using resilience engineering, J. Disaster Res. 12 (4) (2017) 722–732.
- [132] A. Barabadi, Y.Z. Ayele, Post-disaster infrastructure recovery: prediction of recovery rate using historical data, Reliab. Eng. Syst. Saf. 169 (2018) 209–223.
- [133] D. Contreras, G. Forino, T. Blaschke, Measuring the progress of a recovery process after an earthquake: the case of L'Aquila, Italy, Int. J. Disaster Risk Reduct. 28 (2018) 450–464.
- [134] G. Marin, M. Modica, Socio-economic exposure to natural disasters, Environ. Impact Asses. Rev. 64 (2017) 57–66.
- [135] M.A. Mallarangan, M. WasirThalib, G. Yusuf, G.D. Dirawan, The effect of socioekonomic status, environmental knowledge and mitigation attitude toward disaster prevention behavior of community in the coastal area of Makassar city, Int. J. Appl. Environ. Sci. 11 (2) (2016) 637–645.
- [136] D. Sina, A.Y. Chang-Richards, S. Wilkinson, R. Potangaroa, What does the future hold for relocated communities post-disaster? Factors affecting livelihood resilience, Int. J. Disaster Risk Reduct. 34 (2019) 173–183.

- [137] Y. Xu, X. Qiu, X. Yang, X. Lu, G. Chen, Disaster risk management models for rural relocation communities of mountainous southwestern China under the stress of geological disasters, Int. J. Disaster Risk Reduct. (2020) 101697.
- [138] I. Ahmed, T. Gajendran, G. Brewer, K. Maund, J. von Meding, J. MacKee, Compliance to building codes for disaster resilience: Bangladesh and Nepal, Proc. Eng. 212 (2018) 986–993.
- [139] G. Yang, D. Xu, H. Zhang, Catastrophe pre-warning of multi-modular floating platforms with ordinal partition networks, Int. J. Comput. Methods 17 (10) (2020) 2050010.
- [140] M.M. Torok, M. Golparvar-Fard, K.B. Kochersberger, Image-based automated 3d crack detection for post-disaster building assessment, J. Comput. Civ. Eng. 28 (5) (2014) A4014004.
- [141] A.R. Cheema, A. Mehmood, M. Imran, Learning from the past: analysis of disaster management structures, policies and institutions in Pakistan, Disaster Prev. Manag. 25 (4) (2016) 449–463.
- [142] U. Iqbal, J. Barthelemy, W. Li, P. Perez, Automating visual blockage classification of culverts with deep learning, Appl. Sci. 11 (16) (2021).